	GENERAL PUBLIC TARIFF	OF APM TER	MINALS CALLA	0 S.A.							
Item	Services description	Nature	Unit of charge		Tarif	f (USD)			Special	tariff (USD)	v 11.4
SECTION 1		CONTAINERS									
Section 1.1	Standard Services	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
1.1.1	Standard Service to the Vessel - Services to the Vessel										
1.1.1.1	Use or rental of berth (Tariff per hour or fraction hour) (n1)	Regulated	Total LoA (m) x Hours	1.55	0% Rate			1.34	0% Rate		
1.1.2	Loading and Unloading Containers - STS Gantry Cranes (n2 and n3)										
1.1.2.1	Loading and Unloading Full 20 Foot Container with dry cargo		Per move	10.80	0% Rate	194.10	34.94	10.80	0% Rate	173.60	31.25
1.1.2.2	Loading and Unloading Full 40 Foot Container with dry cargo		Per move	10.80	0% Rate	294.10	52.94	10.80	0% Rate	263.60	47.45
1.1.2.3	Loading and Unloading Full 20 Foot Container with refrigerated cargo		Per move	10.80	0% Rate	194.10	34.94				
1.1.2.4	Loading and Unloading Full 40 Foot Container with refrigerated cargo		Per move	10.80	0% Rate	294.10	52.94				
1.1.2.5	Loading and Unloading Empty 20 Foot Container	Regulated	Per move	151.10	0% Rate			131.10	0% Rate		
1.1.2.6	Loading and Unloading Empty 40 Foot Container		Per move	235.50	0% Rate			193.70	0% Rate		
1.1.2.7	Complete cycle 20 Foot Transhipment Container (full or empty) within the Terminal (n4)		Per complete cycle	184.00	0% Rate			119.00	0% Rate		
1.1.2.8	Complete cycle 40 Foot Transhipment Container (full or empty) within the Terminal (n4)		Per complete cycle	282.00	0% Rate			178.00	0% Rate		
1.1.3	Loading and Unloading Containers - Without STS Gantry Cranes (n3 and n5)										
1.1.3.1	Loading and Unloading Full 20 Foot Container with dry cargo		Per move	5.90	0% Rate	176.40	31.75	5.90	0% Rate	139.90	25.18
1.1.3.2	Loading and Unloading Full 40 Foot Container with dry cargo		Per move	5.90	0% Rate	268.10	48.26	5.90	0% Rate	213.30	38.39
1.1.3.3	Loading and Unloading Full 20 Foot Container with refrigerated cargo		Per move	5.90	0% Rate	176.40	31.75				
1.1.3.4	Loading and Unloading Full 40 Foot Container with refrigerated cargo	Description	Per move	5.90	0% Rate	268.10	48.26				
1.1.3.5	Loading and Unloading Empty 20 Foot Container	Regulated	Per move	125.00	0% Rate			111.70	0% Rate		
1.1.3.6	Loading and Unloading Empty 40 Foot Container		Per move	195.10	0% Rate			166.20	0% Rate		
1127	Complete curls 20 East Transhipment Continour (full or complet within the Terminal (n/l)		Por complete cuelo	147.00	09/ Pato			100.00	09/ Poto		

1.1.3.2 Coordinate and Unicodating Empty 40 Food Container of Container of Land Container of Container of Land Container of Container of Land Container of L

Section 1.2	Special Services to the Vessel (n6)	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff) Cargo (IG	<ul> <li>Vessel (Tariff)</li> </ul>	Vessel (IGV)	Cargo (Tariff) Cargo
1.2.1	Special Services to the Vessel (hb)  Hatches or Hold Lids (n7)	nature	Onit of charge	resser (Tarrit)	vesser (IOV)	Cargo (Tantil) Cargo (IG	yesser (rann)	vesser (IGV)	Cargo (Tariff) Cargo
1.2.1.1	Movement of ISO hatches	Regulated	Per move	0.00	0% Rate				
1.2.2	Restow of containers (n6)	Regulated	Pel move	0.00	Una Mate				
1.2.2.1	Restow are hold 20 foot standard container		Per container	0.00	0% Rate				
1.2.2.1	Restow same noid zu root standard container  Restow same hold 40 foot standard container		Per container Per container	0.00	0% Rate				
1.2.2.3	Restow via guay 20 foot standard container	Regulated	Per container	0.00	0% Rate				
1224	Restow via guay 40 foot standard container		Per container	0.00	0% Rate				
1.2.3	Increase of container productivity with Mobile Harbor Cranes (MHC)		Per container	0.00	Un Male				
1.2.3.1	Supply of MHC to increase productivity during loading/unloading container operations (price per shift or shift fraction) (n8)	Non Regulated	Per Crane/Shift	2,500.00	0% Rate				
1.2.4	Use of Operational Area - Import Empty Containers (except Transhipment)	Non Regulated	rei Gialleidillit	2,300.00	U /s INdice				
1.2.4.1	48 hours (Free time included in Standard Service) (n9)	Regulated		Free					
1.2.4.2	Day: 3 – 6 (Tariff per day)	Neguateu	Per TEU/day	10.70	0% Rate				
				37.50					
1.2.4.3	Day 7 (Price per day)	Non Regulated	Per TEU/day		0% Rate				
1.2.4.4	Day: 8 onwards (Price per day)		Per TEU/day	16.10	0% Rate				
1.2.5	Use of Operational Area - Export Empty Containers (except Transhipment)								
1.2.5.1	72 hours (Free time) (n10)	Regulated		Free					
1.2.5.2	Days: 4 – 6 (Tariff per day)		Per TEU/day	10.70	0% Rate				
1.2.5.3	Day 7 (Price per day)	Non Regulated	Per TEU/day	37.50	0% Rate				
1.2.5.4	Day: 8 onwards (Price per day)		Per TEU/day	16.10	0% Rate				
1.2.6	Use of Operational Area - Transhipment Full Containers								
1.2.6.1	48 hours (Free time included in Standard Service) (n11)	Regulated		Free	e				
1.2.6.2	Day: 3 - 4 (Price per day)		Per TEU/day	17.10	0% Rate				
1.2.6.3	Day: 5 – 6 (Price per day)	Non Regulated	Per TEU/day	26.80	0% Rate				
1.2.6.4	Day 7 onwards (Price per day)	real regulated	Per TEU/day	33.00	0% Rate				
1.2.7	Use of Operational Area - Transhipment Empty Containers		rei iLoiday	33.00	U /s INdice				
				Free					
1.2.7.1	48 hours (Free time included in Standard Service) (n11)	Regulated							
1.2.7.2	Day: 3 – 4 (Price per day)		Per TEU/day	17.10	0% Rate				
1.2.7.3	Day: 5 – 6 (Price per day)	Non Regulated	Per TEU/day	26.80	0% Rate				
1.2.7.4	Day 7 onwards (Price per day)		Per TEU/day	33.00	0% Rate				
1.2.8	Use of Operational Area for Empty pool - Empty containers (except Transhipment)								
1.2.8.1	Use of operational area for empty containers from empty pool (n12)	Non Regulated	Per TEU	Per agre	ement				
1.2.9	Special Container - empty and transhipment								
1.2.9.1	Loading/unloading and delivery of non ISO/OOG (Additional) Container 20 and 40 foot (n7)	Regulated	Per container	0.00	0% Rate				
1.2.9.2	Treatment of Hazardous cargo containers IMO class 1 (n13)		Per TEU	260.00	0% Rate				
1.2.9.3	Treatment of Hazardous cargo containers IMO class 5.2 class 6.2 and class 7 (n13)		Per TEU	260.00	0% Rate				
1.2.9.4	Treatment of Hazardous cargo IMO 9 (n13)		Per TEU	120.00	0% Rate				
1.2.9.5	Treatment of Hazardous cargo other IMO classes (n13)		Per TEU	150.00	0% Rate				
1.2.9.6	Supply of special equipment for handling 20' containers with oversized cargo in the container yard	Non Regulated	Container-Move	169.00	0% Rate				
1.2.9.7	Supply of special equipment for handling 20 containers with oversized cargo in the container yard		Container-Move	330.00	0% Rate				
				25.00					
1.2.9.8	Containers with oversized cargo - use of operational area (n14)		Per container/day		0% Rate				
1.2.9.9	High Cube Containers 9'6" (n14)		Per container	20.60	0% Rate				
1.2.10	Complementary Services to stevedoring/unstevedoring								
1.2.10.1	Connection and disconnection service for reefer containers on-board vessels (n15)	Non Regulated	Per container-operation	9.00	0% Rate				
1.2.11	Transhipment Reefer Containers								
1.2.11.1	Energy - container 20 and 40 foot (n6 y n16)	Regulated	Per Container/hour	0.00	0% Rate				
1.2.11.2	Inspection and monitoring - container 20 and 40 foot (n7 and n16)	Regulated	Per Container/day	0.00	0% Rate				
1.2.11.3	Energy - container 20 and 40 foot (n17)	Non Regulated	Per Container/day	75.00	0% Rate				
1.2.11.4	Inspection and monitoring - container 20 and 40 foot (n17)	ivuii regulated	Per event	27.20	0% Rate				
1.2.12	Re-stow Reefer Containers (n18)								
1.2.12.1	Energy supply for re-stow reefer containers	Non Regulated	Per container	77.10	0% Rate				
1.2.13	Gasification of a full container reefer - all traffics (n19)								
1.2.13.1	Supply of gases N2 y CO2		Per container	503.30	90.59		480.00	86.40	
1.2.13.1	Supply of gases N2	Regulated	Per container	440.00	79.20		420.00	75.60	
		regulated		105.00	18.90		920.00	75.00	
1.2.13.3	Supply of gases CO2		Per container	105.00	18.90				
1.2.14	Additional inspection of seals								
1.2.14.1	Additional inspection or revision of seals (n20)	Non Regulated	Per seal	12.00	0% Rate				
1.2.15	Traction Service for Inter-terminal Transhipment Containers								
1.2.15.1	Inter-terminal Transport of Transhipment Containers (n21)	Non Regulated	Per container	40.00	0% Rate				

1.2.1. Inter-terminal Transport of Transport

(n/4) This expirate ones additional departs of continuency safet or departs of states of entity containers gain attention that is starting restored. The containers are containers are containers are containers are containers are containers.

(n/5) This applicable to all discharged or shipped containers through the Terminal including restore containers (n/4 quay and in the same hold), that require such service. This is applicable for this operation, whether it is connection or decorrection.

(n/17) This applicable to this 17th disposance.

(n/17) This applicable to the 17th disposance.

(n/17) This applicable to the 17th disposance.

(n/17) This applicable to the 17th disposance.

(n/18) This applicable to the 17th disposance.

(n/18) This applicable to the 17th disposance.

(n/18) This applicable to the settlem containers are quarter to a quarter to the containers are the settlem of the settlem

Section 1.3	Special Service of Use of Operational Area (Port Terminal) - Services to the Cargo	Nature	Unit of charge	Vessel (Tariff) Vess	el (IGV) Cargo (Tariff	Cargo (IGV)	Vessel (Tariff) Vessel (IGV) Cargo (Tariff) Cargo (IG	V)
1.3.1	Use of Operational Area - Import Full Containers (except Transhipment) (n22)							
1.3.1.1	48 hours to import and/or discharge container (Free time included in Standard Service) (n23)	Regulated				ree		
1.3.1.2	Day: 3 – 6 (Tariff per day)		Per TEU/day		40.0			
1.3.1.3	Day 7 (Price per day) container 20 and 40 foot		Por TEU		85.0			
1.3.1.4	Days: 8 – 10 (Price per day)	Non Regulated	Per TEU/day		42.0			
1.3.1.5	Days: 11 – 20 (Price per day)	Ivoii ivegulateu	Per TEU/day		42.0			
1.3.1.6	Days: 21 – 28 (Price per day)		Per TEU/day		42.0			
1.3.1.7	Day: 29 onwards (Price per day)		Per TEU/day		42.0	7.56		
1.3.2	Use of Operational Area - Export Full Containers (except Transhipment) (n22)							
1.3.2.1	72 hours to export and/or loading container (Free time) (n24)	Regulated				ree		
1.3.2.2	Days: 4 – 6 (Tariff per day)		Per TEU/day		37.5			
1.3.2.3	Day 7 (Price per day) container 20 and 40 foot		Por TEU		85.0			
1.3.2.4	Days: 8 – 10 (Price per day)	Non Regulated	Per TEU/day		42.0			
1.3.2.5	Days: 11 – 20 (Price per day)	Ivoit Negulateu	Per TEU/day		42.0			
1.3.2.6	Days: 21 – 28 (Price per day)		Per TEU/day		42.0			
1.3.2.7	Day: 29 onwards (Price per day)		Per TEU/day		42.0	7.56		

prices escapatione in the Contessant Limitation.

In this day, the contessant Limitation is a withdrawn from the yard. Fire firm does not apply to full transhipment containers with oversized cargo, that is, for this type of container, the use of operating seas is bitled from the first day, urapply ends with vesser's Estimated Firm of Berthing (ETIS) to be communicated to users. For purposes of computation of discage fee time storage and billable storage, it should be balen in consideration in the so-called "Sallable ETIS" that is published in the total transhipment containers with revealed cargo, but it is, for this for deprine desirable the users of the container. The use of operating seas is bitled from the first day.

In the container is the container that the user of the container with overall container. The user of operating seas is bitled from the first day.

In this day to the container, the user of operating seas is bitled from the first day.

Section 1.4	Temporary Depot Special Service - Services to the Cargo	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff) Vessel (IGV) Cargo (Tariff)	Cargo (IC
1.4.1	Temporary Depot Integrated Service for dry import containers (n25)								
	20' Container - Includes use of area until the 10th day		Per container			302.10	54.38		
	40' Container - Includes use of area until the 10th day		Per container			399.60	71.93		
	Use of area during days 11 - 20 (Price per day) (n26)	Non Regulated	Per TEU/day			15.00	2.70		
	Use of area during days 21 – 28 (Price per day) (n26)		Per TEU/day			20.00	3.60		
	Use of area from 29th day onwards and other services (n26)			See section 1.3.1	7 and others from	section 1.5			
	Temporary Depot Integrated Service for dry export containers (n27)								
	20' Container - Includes use of area until the 7th day		Per container			243.10	43.76		
	40' Container - Includes use of area until the 7th day		Per container			339.60	61.13		
	Use of area days 8 – 10 (Price per day) (n28)	Non Regulated	Per TEU/day			7.00	1.26		
	Use of area days 11 – 20 (Price per day) (n28)	real regulated	Per TEU/day			9.00	1.62		
	Use of area during days 21 – 28 (Price per day) (n28)		Per TEU/day			20.00	3.60		
	Use of area from 29th day onwards and other services (n28)			See section 1.3.2	7 and others from	section 1.5			
	Temporary Depot integrated service for reefer import containers (n29)								
	20' Container - Includes use of area until the 7th		Per container			374.60	67.43		
	40' Container - Includes use of area until the 7th	Non Regulated	Per container			463.50	83.43		
	Use of area during days 8 – 10 (Price per day) (n30)		Per TEU/day			10.00	1.80		
	Use of area from day 11 onwards and other services (n30)		See se	ction 1.4.1.3, 1.4.1.4	and 1.3.1.7 and o	thers from section 1.	5		
	Temporary Depot integrated service for reefer export containers (n31)								
	20' Container - Includes use of area until the 7th		Per Container			341.60	61.49		
	40' Container - Includes use of area until the 7th	Non Regulated	Per Container			411.50	74.07		
1443	lice of area from day 8 nawards and other services (n32)		See	parting 1 4 2 3 to 1 4	25 1327 and c	there of section 1.5			

1.4.3 Use of area form 6.8 2 museful and die 2 museful and die 3 m

(1/2) This shall be applicable sherever APM Termins Collato S.A. has been nominated as Temporary Depot (1011), Service includes Cappage days.

(1/2) This shall be applicable sherever APM Termins Collato S.A. has been nominated as Temporary Depot (1011), Service includes Cappage days.

(1/2) This shall be applicable sherever APM Termins Collato S.A. has been nominated as Temporary Depot (1011), Service includes Cappage days.

(1/2) This shall be applicable sherever APM Termins Collato S.A. has been nominated as Temporary Depot (1011), Service includes Cappage days.

(1/2) This shall be applicable to make of present or a service of the continuer is withboard service.

(1/2) This shall be applicable to open of present or a service of the continuer is withboard service.

(1/2) This shall be applicable to open of present or a service of the continuer is withboard service.

(1/2) This shall be applicable to open of present or a service of the continuer is withboard service.

(1/2) This shall be applicable to open of present or a service of the continuer is withboard service.

(1/2) This shall be applicable to open of present or a service of the continuer is withboard service.

(1/2) This shall be applicable to open of present or a service of the continuer is withboard service.

(1/2) This shall be applicable to open of present or a service of the continuer is withboard service.

(1/2) This shall be applicable to open of present or a service of the continuer is withboard service.

(1/2) This shall be applicable to open of the continuer is withboard service.

(1/2) This shall be applicable to open of the continuer is withboard service.

(1/2) This shall be applicable to open of the continuer is withboard service.

(1/2) This shall be applicable to open of the continuer is withboard service.

(1/2) This shall be applicable to open of the continuer is withboard service.

(1/2) This shall be applicable to open of the continuer is withboard service.

(1/2) This shall be applicable to open of the continuer is withboard serv

(I)(22) Price of section 1.4.2.3 is applicable to use of operational was form days 8 to 10 of Temporary Depty price of section 1.5.2.4 is applicable for days 21 to 20), and price of section 1.3.2.7 is applicable for days 22 onwards. Price of section 1.5.2.3 is applicable for energy supply for day 8th onwards and price of section 1.5.2.4 is applicable for days 22 onwards. Price of section 1.5.2.3 is applicable for energy supply for day 8th onwards and price of section 1.5.2.4 is applicable for days 22 onwards. Price of section 1.5.2.3 is applicable for energy supply for day 8th onwards and price of section 1.5.2.3 is applicable for energy supply for day 8th onwards and price of section 1.5.2.3 is applicable for energy supply for day 8th onwards and price of section 1.5.2.3 is applicable for energy supply for day 8th onwards and price of section 1.5.2.3 is applicable for energy supply for day 8th onwards and price of section 1.5.2.3 is applicable for energy supply for day 8th onwards and price of section 1.5.2.3 is applicable for energy supply for day 8th onwards and price of section 1.5.2.3 is applicable for energy supply for day 8th onwards and price of section 1.5.2.3 is applicable for energy supply for day 8th onwards and price of section 1.5.2.3 is applicable for energy supply for day 8th onwards and price of section 1.5.2.3 is applicable for energy supply for day 8th onwards and price of section 1.5.2.3 is applicable for energy supply for day 8th onwards and price of section 1.5.2.3 is applicable for energy supply for day 8th onwards and price of section 1.5.2.3 is applicable for energy supply for day 8th onwards and price of section 1.5.2.3 is applicable for energy supply for day 8th onwards and price of section 1.5.2.3 is applicable for energy supply for day 8th onwards and price of section 1.5.2.3 is applicable for energy supply for day 8th onwards and price of section 1.5.2.3 is applicable for energy supply for day 8th onwards and price of section 1.5.2.3 is applicable for energy supply for day 8th onwa

Section 1.5	Other Special Services for Containers (Port Terminal & Tempory Depot) - Services to the Cargo	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	
		resture	Offic of charge	vesser (Tallill)	vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	à
1.5.1	Special Cargo		D. TEU			000.00	10.00	
1.5.1.1	Treatment of Hazardous cargo containers IMO class 1 (n33		Per TEU			260.00	46.80	
1.5.1.2	Treatment of Hazardous cargo containers IMO class 5.2, class 6.2 and class 7 (n33)		Per TEU			260.00	46.80	
1.5.1.3	Treatment of Hazardous cargo IMO 9 (n33 and n34)		Per TEU			120.00	21.60	
1.5.1.4	Treatment of Hazardous cargo other IMO classes (n33)	Non Regulated	Per TEU			150.00	27.00	
1.5.1.5	Supply of special equipment for handling 20' containers with oversized cargo in the container yard		Container-Move			169.00	30.42	
1.5.1.6	Supply of special equipment for handling 40' containers with oversized cargo in the container yard		Container-Move			330.00	59.40	
1.5.1.7	Containers with oversized cargo - use of operational area (n35)		Per container/day			25.00	4.50	
1.5.1.8	High Cube Containers 9'6" (n35)		Container			20.60	3.71	
1.5.2	Reefer Container							
1.5.2.1	Energy - 20' and 40' container (n36 and n37)	Regulated	Per container/hour			0.00	0.00	
1.5.2.2	Inspection and monitoring - 20' and 40' container (n36 and n37)	rtegulateu	Per container/day			0.00	0.00	
1.5.2.3	Energy - 20' and 40' container (Price per day or fraction day) (n.38)		Per container/day			75.00	13.50	
1.5.2.4	Inspection and monitoring - 20' and 40' container (n38)	Non Regulated	Per event			27.20	4.90	
1.5.2.5	Pre-cooling empty container (n39)	rvon rteguateu	Per Container			36.00	6.48	
1.5.2.6	Provision of cold antechamber (n40)		Per Unit/day			250.00	45.00	
1.5.2.7	Pre-cooling empty container (n39)		Per Container			36.00	6.48	
1.5.2.8	Assembly and disassembly of clip system on reefer Gensets (n36)	Regulated	Per complete cycle			0.00	0.00	
1.5.3	Cargo Handling and horizontal movement operations					0.00		
1.5.3.1	Additional Movement to Standard Service in the terminal as per request of the user or authorities (n36)		Per Container			0.00	0.00	
1.5.3.2	Cargo inspection with groups o gangs - 20' and 40' container (n36)	Regulated	Per Container			0.00	0.00	
1.5.3.3	Cargo inspection with forkifts - 20' and 40' container (n36)		Per Container			0.00	0.00	
1.5.3.4	Consolidation/deconsolidation of containers with dry caroo (with forkift) (n36)		Per Container			176.00	31.68	
1.5.3.5	Consolidation/deconsolidation of containers with dry cargo (with crew or crew-forkiff) (n36)		Per Container			200.00	36.00	
1.5.3.6	Consolidation/deconsolidation of containers with dry cargo (with coil handling forkiff or another type of special forkiff) (n36)		Per Container			230.00	41.40	
1.5.3.7	Consolidation (econsolidation of containers with dry cargo (with forhiff) including Gate In (Gate Out of empty container (n/36)		Per Container			230.00	41.40	
1.5.3.8	Consolidation/deconsolidation of containers with dry cargo (with rown) including deal in Cate of the Impact of the		Per Container			250.00	45.00	
1.5.3.9	Consolidation/deconsolidation of containers with dry cargo (with cell handling forkiff or another type of special forkiff) including Gate Intigate Out of empty containers (150)	Non Regulated	Per Container			280.00	50.40	
1.5.3.10	Consolidation deconsolidation of containers with refrigerated cargo.	Ivoii Negulateu	Per Container			660.00	118.80	
1.5.3.10	Consolidation/deconsolidation of containers with refrigerated cargo including Gate In/Gate Out of emoty containers		Per Container			869.00	156.42	
1.5.3.11			Per Container Per shift			1,250.00	156.42 225.00	
1.5.3.12	Consolidation/deconsolidation of containers with Mobile Harbour Cranes (Price per shift or fraction shift) (n41)  Partial consolidation/deconsolidation of containers with dry cargo		Per Container			1,250.00	18.00	
	Partial consolidation deconsolidation of containers with dry cargo  Depalletizing (removal of straps and/or strapping) (n42)		Per Container Per Pallet			100.00	18.00	
1.5.3.14	Depaietzing/paletzing (removal or straps and/or strapping) (n42)  Other Services		rei Pallet			10.00	1.80	
1.5.4.1	Utnet services Breakdown as per instructions of freight forward agent. (n43)		Per B/I			23.00	4.14	
			Per B/L Per breakdown			23.00		
1.5.4.2	Bill of lading breakdown (n43)						4.14	
1.5.4.3	Sweeping of empty container		Per container			6.00	1.08	
1.5.4.4	Simple cleaning of container		Per container			24.00 48.00	4.32 8.64	
1.5.4.5	Chemical cleaning of container	Non Regulated	Per container					
1.5.4.6	Container repair		Per Activity			Per c		
1.5.4.7	Seal placement services		Per seal			12.00	2.16	
1.5.4.8	Additional inspection or revision of seals (n20)		Per seal			12.00	2.16	
1.5.4.9	Labels Placing or Removal Services (n44)		Per container			9.00	1.62	
1.5.4.10	Additional weighing of containers (n45)		Per container			90.00	16.20	
1.5.5	Depot for Empty Containers							
1.5.5.1	Gate In (n46)	Non Regulated	Per container			240.00	43.20	
1.5.5.2	Gate Out (n46)	guutou	Per container			170.00	30.60	

1.5.5.2 Gate Dut (n6%)
1.5.5.3 Gate Dut (n6%)

(no.) The service covers additional copyoid or continuery safet or stacks of MLI continuers (seed or 14, services). In the case of continuers with oversized carge, the use of leas is billed from the first day (does not include fee (miles). Applicable to the services are becaused; responsible to proceed the expectation of the services. (10.0) This is applicable to all continuers are processed or services are services. (10.0) This is applicable to all continuers. (10.0) This is applicable to all continuers are processed or services are processed or services. (10.0) This is applicable to may of reservice for the first Placebaldon on the first Reservices are continued as in responsible from the first Reservices are continued as in responsible from the first Reservices are continued as applicable to may of reservice for the first Reservices are continued as applicable to may of reservice for the first Reservices are continued as applicable from the first Reservices are continued as applications to the first Reservices are continued as applications to a first Reservices are continued as applications to the first Reservices are continued as a continued as applications to the first Reservices are continued as applications to the first Reservices are continued as applications to the first Reservices are continued as applications to

(AC) It is charged to each breakdown BL mother SL sour or BL grandson.

(AC) Find purple to each breakdown BL mother SL sour or BL grandson.

(AC) Report of the state of the

SECTION 2	BF	REAK BULK CARGO S	ERVICES								
Section 2.1	Standard Services	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
2.1.1	Standard Service to the Vessel - Services to the Vessel										
2.1.1.1	Use or rental of berth (Tariff per hour or fraction hour) (n1)	Regulated	Total LoA (m) x Hours	1.55	0% Rate			1.34	Tasa 0%		
2.1.2	Standard Service to Break Bulk Cargo										
2.1.2.1	Loading and Unloading of Break Bulk cargo (n47)	Regulated	Per Ton	2.50	0% Rate	14.90	2.68	2.50	0% Rate	13.40	2.41
(n47): Vessel port	on of this tariff includes transference of the cargo from vessel to the quay, lashing/unlashing and tally. Payment of Vessel portion will be applicable to the shipping line or cargo consignee, up	on conditions set forth in	the maritime shipping contrac	. If Vessel portion of t	his tariff is charged to	o the cargo consignee	, then IGV (IVA) shall	be applicable to this	tariff.		
Section 2.2	Special Services - Services to the Vessel (n6)	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
2.2.1	Hatches or Hold Lids (n7)										
2.2.1.1	Movement of ISO hatches	Regulated	Per move	0.00	0% Rate						
2.2.2	Re-stow of Break Bulk Cargo										
2.2.2.1	Re-stow via vessel	Regulated	Per ton	17.20	0% Rate			16.00	Tasa 0%		
2.2.2.2	Re-stow via quay (n48)		Per ton	34.70	0% Rate			33.00	Tasa 0%		
2.2.3	Transhipment of Break Bulk cargo			05.00	00 D :						
2.2.3.1	Transhipment of Break Bulk cargo - Complete cycle (n49)  [cable to the cargo that is re-stowed in the same hold flevell or in another hold flevell of the vessel which in both cases cargo requires to be temporarily transferred to the quay.	Non Regulated	Per ton	25.60	0% Rate						
	icacion to the cargo trist is re-stowed in the same notic (level) of in another hold (level) of the vessel which in coin cases cargo requires to be temporarily transferred to the quay, icle includes 3 fixed days of storage which applies after completion of cargo discharge operation of the vessel.										
Section 2.3	Special Service of Use of Operational Area (Port Terminal) - Services to the Cargo	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
2.3.1	Use of Operational Area - All traffic										
2.3.1.1	Days: 1 - 3 (Free time - included in Standard Service)	Regulated				Fn					
2.3.1.2	Days: 4 – 10 (Tariff for whole period or fraction period)		Per ton			26.00	4.68				
2.3.1.3	Days: 11 - 20 (Tariff per day or fraction day)	Non Regulated	Per ton/day			2.90	0.52				
2.3.1.4	Days: 21 - onwards (Tariff per day or fraction day)		Per ton/day			4.20	0.76				
Section 2.4	Special Service of Temporary Depot - Services to the Cargo	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
2.4.1	Temporary Depot Integrated Service - All traffic (n50)										
2.4.1.1	Break Bulk Cargo - Includes use of operational area until day 10		Per ton			29.00	5.22				
2.4.1.2	Use of operational area from days 11 - 20 (Price per day of day fraction)		Per ton/day			1.50	0.27				
2.4.1.3	Use of operational area from day 21 onwards	Non Regulated	Per ton/day			4.20	0.76				
2.4.1.4	Break Bulk Cargo - Includes use of operational area until day 30 (n51)		Per ton			70.00	12.60				
2.4.1.5	Other services (n52)				2.3.1.4 and 2.5.2 to						
(n50): This is appl	cable whenever APM Terminals Callao S.A. is appointed as Temporary Depot (code 3014) after cargo has been previously coordinated and accepted by APMTC (Subject to space availability	y). It includes the Standan	d Service cargo portion, use o	f operational area unt	il day 10, temporary	depot documentation	emission (volante) ar	d transmission to Co	ustoms. First three o	alendar days are free	of charge and are

part of the Standard Service.

(n51) Price of section 2.4.1.3 is applicable for use of operational area of temporary deposit of subsection 2.4.1.4 from day 31 and onwards.

(nS2): For the other services, tariffs and prices of sections from 2.5.2 to 2.5.4 are applicable independently from the number of storage days of break bulk cargo in the Terminal and regardless of the temporary deposit services chosen (either from section 2.4.1.1 or 2.4.1.4)

Other Special Services to Break Bulk Cargo (Port Terminal & Temporary Depot) - Services to the Cargo	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff) V	essel (IGV)	Cargo (Tariff)	Cargo (IGV)
Use of Simple Depot - All traffic (n53)										
Days: 1 - 30 (Price per month or fraction month)	Non Regulated	Per m2			11.00	1.98				
		Per ton or m3				5.76				5.36
	Dogulated	Per ton or m3				8.03			41.20	7.42
	rveguateu	Per ton								
		Per ton			23.40	4.21				
		Per ton			4.00	0.72				
Handling for registry of break bulk cargo (n56)										
	Non Regulated	Per MAFI			281.00					
	Ivoii ivegulateu	Per Shift			1,250.00	225.00				
		Per MAFI	83.00	14.94						
Bagging or racking of goods in sacks or big bags (60)		Per ton			20.00	3.60				
Other services										
Additional weighing (This does not include extra movement)	Non Regulated	Per truck			21.00	3.78				
	Jace of Simple Deport - All traffic (nSS) Jaces 1-30 (Piche per mother fraction month) Special Cargo Assing Uniosating of project cargo without mobile harbor crane (nS4) Jacking Uniosating of project cargo without mobile harbor crane (nS4) Jacking Uniosating of project cargo with mobile harbor crane (nS4) Jacking Uniosating of project cargo with mobile harbor crane (nS4) Jacking Uniosating Cargo (Laboration cargo uniosating Cargo (Laboration Cargo (Laboratio	Ixe of Simple Deport - All traffic (nS)  spec 1-30 (Pice per mother fraction month)  Non Regulated Special Cargo  acting Unitediate of project cargo without mobile harbor crane (nS4)  acting Unitediate of project cargo without mobile harbor crane (nS4)  acting Unitediate of project cargo without mobile harbor crane (nS4)  acting Unitediate of project cargo without mobile harbor crane (nS4)  acting Unitediate of project cargo with mobile harbor crane (nS4)  acting Unitediate of project cargo with mobile harbor crane (nS4)  acting Unitediated on Cargo Unitediated Special Cargo Unitediated Deportment of the Special Cargo Unitediated Special Cargo Unitediated Deportment of the Special Cargo Unitediated Deportment of Cargo Unitediated Deportment of Unitediated Depo	Non Regulated   Per m2	See of Simple Deport All Intiffic (rick)	Non Regulated   Per m2	Isse of Simple Deport - All traffic (rich3)  special Cargo get 1.3 (Pitce per month or fraction month)  Non Regulated Per m 2  11.00 percial Cargo get of Simple Cargo without mobile harbor crarse (ric54)  acting Unicidating of project cargo without mobile harbor crarse (ric54)  Acting Unicidating of project cargo without mobile harbor crarse (ric54)  Acting Unicidating of project cargo without mobile harbor crarse (ric54)  Acting Unicidating of project cargo without mobile harbor crarse (ric54)  Acting Unicidating of project cargo without mobile harbor crarse (ric54)  Acting Unicidating of Per ton or m3  Acting Unicidating of Cargo per Cargo Harborito Simple Cargo Per ton 023.40  Acting Harborito Simple Cargo Per ton 04.00  Acting Fragility of treat butk cargo (ric5)  Acting Fragility of treat butk cargo (ric5)  Per ton 04.00  Acconsolidation Consolidation of brank butk cargo (ric5)  Per Shift Per Shift Per Shift Acting Cargo Harborito (rich Shift) Per Shift Acting Cargo Harborito (rich Shift) Per Shift Acting Cargo (rich	See of Simple Deport All Intellif (ridS)   Signet 1-30 (Pines premit or Faction month)   198   Special Cargo   198   Special Cargo   198   Special Cargo   198   Per ton or m3   200   5.76   Additional Control Cargo without mobile harbor caree (rid5)   88   Per ton or m3   44.60   8.00   Additional Control Cargo without mobile harbor cargo (rid5)   88   Regulated   Per ton or m3   44.60   8.00   Additional Control Cargo without mobile harbor cargo (rid5)   89   Per ton or m3   44.60   8.00   Additional Control Cargo without mobile harbor cargo (rid5)   89   Per ton or m3   44.60   8.00   Additional Control Cargo without mobile harbor cargo (rid5)   89   Per ton or m3   44.60   8.00   Additional Control Cargo without mobile harbor cargo (rid5)   89   Per ton or m3   44.60   8.00   Additional Control Cargo without mobile harbor cargo (rid5)   89   Per ton or m3   44.60   8.00   Additional Control Cargo without mobile harbor cargo (rid5)   89   Per ton or m3   44.60   8.00   Additional Control Cargo with Marks (rid5)   89   Per ton or m3   44.60   4.00   Additional Cargo with Marks (rid5)   89   Per ton or m3   4.00   0.77   Per ton or m3   0.00	See of Simple Deport - All traffic (rich3)	See of Simple Deport - All traffic (n/S)	Non Regulated   Per m2   1100   158

ins 3.7 This is applicable but careing the Black carego, activing in a designated was not stronging if for a protect gar importer our importer and APM Terminals Callas S.A. This is applicable for Bleak But carego half will be finally shipped as another type of carego (in containers mainly) or for another carego that has been unloaded in caller But Arm S.A. This is applicable for Bleak But carego half carego half are been been been unloaded in a Bleak But carego in this cases. I staff for this service is applicable as a simple begat only. Space is subject to availability.

(incl.) This a applicable but all cape that cannot be betteringer by the carego that is not carego containers mainly or or short one or being one 30 mJ per unit. The unit of charge shall be applicable to all cape that cannot be bettering to the shipping line or carego consigners excording to the martine shipping contract.

(n55) Twiff corresponds to loading and unloading operations of hazardous break bulk cargo (MID) wherever additional equipment antistr personnel is to be provided as per Port Authority regulation or upon request of the user. Twiffs of sections 2.5.23 and 2.5.24 include Standard Service to break bulk cargo. If additional equipment and perspectively regulation or upon request of the user. Twiffs of sections 2.5.23 and 2.5.24 include Standard Service to break bulk cargo. If additional equipment and perspectively regulation or upon request of the user. Twiffs of sections 2.5.23 and 2.5.24 include Standard Service to break bulk cargo. If additional equipment and perspectively regulation or upon request of the user. Twiffs of sections 2.5.23 and 2.5.24 include Standard Service to break bulk cargo. If additional equipment and perspectively regulated in the user of the storage area.

in(5). This eless to handling of carpy using brilling and/or gauge for the registry or inspection of carpy. This service shall be provided upon request of the user (usage conner or consigner) or the subtraffee.

in(5) Price as splicable for special between to break but cargo in the storage year in order to be separatellisposthyrization of price-badd break but cargo in cargo for metalic postures (usage for metalic postures (usage for metalic postures). This is mixed to the cargo consigner.

in(5) Service include the use of black in Harbor or of desset but cargo in order to be separatellished postures are precised as the cargo consigner.

in(6) Service include the use of black in Harbor or of desset but cargo for metalic postures (usage for metalic postures). The service is not order to be cargo consigner, being applicable in the cases when Mobile Harbor Oranes are required for res than four hour (u).

(mid) defined features are used in security to the contract of the contract of

SECTION 3		RO-RO CARGO SER	VICES								
Section 3.1	Standard Service	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
3.1.1	Standard Service to the Vessel - Services to the Vessel										
3.1.1.1	Use or rental of berth (Tariff per hour or fraction hour) (n1)	Regulated	Total LoA (m) x Hours	1.55	0% Rate			1.34	Tasa 0%		
3.1.2	Standard Service to Ro-Ro Cargo		Perton	3.50	0% Rate	44 10	7 94	3.50	0% Rate	42.80	7.70
3.1.2.1	Loading or Unloading of Ro-Ro Cargo (n61)  flor of this tariff will be charged to the Shipping line (or representative), unless maritime shipping contract indicates otherwise. If Vessel portion of this tariff is charged to the cargo consignee.	Regulated		3.50	U% Rate	44.10	7.94	3.50	U% Rate	42.80	7.70
(no i): vessei por	auth or this tariff will be charged to the Shipping line (or representative), unless manufine shipping contract indicates otherwise, it vesses ponton or this tariff is charged to the cargo consignee, it	nen IGV (IVA) shall be ap	sicable to this tanii.								
Section 3.2	Special Services - Services to the Vessel	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
3.2.1	Re-stow of Ro-Ro cargo										
3.2.1.1	Re-stow via vessel	Regulated	Per ton	52.80	0% Rate			50.30	Tasa 0%		
3.2.1.2 3.2.2	Re-stow via quay (n62) Transhioment of Ro-Ro cargo		Per ton	102.90	0% Rate			99.00	Tasa 0%		
3.2.2	Transhipment of No-Ro cargo - Complete cycle (n63)	Non Regulated	Per ton	72.80	0% Rate						
3.2.2.1	Transipprier or ro-ro dago - Compete cycle (165) Use of Operational Area - Transipment (166)	Ivon requisied	Per ton	12.00	Un Male						
3.2.3	Use of Operational Area - Transmipment (not)  Dias: 1-3 (Tiempo libre - included in the special service ro-ro caroo transhipment)			Fr	ree						
3232	Dias: 1-5 (Heinplanie - Induced in this special service root daily transmipment) Days: 4 - 10 (Price for total period or fraction period)	No Regulado	Per ton	48.80	Tasa 0%						
3.2.3.3	Days: 11 - onwards (Price per day or fraction day)		Per ton/day	20.00	Tasa 0%						
(n62): This is app	plicable to the cargo that is re-stowed in the same hold (lievell) or in another hold (levell) of the vessel which in both cases cargo requires to be temporarily transferred to the quay.										
	rvice includes 3 free days of storage which applies after completion of cargo discharge operation of the vessel.										
(n64): Prices of t	his section are applicable to all types of vehicle and machinery.										
Section 3.3	Special Service of Use of Operational Area (Port Terminal) - Services to the Cargo	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
Section 3.3 3.3.1	Special Service of Use of Operational Area (Port Terminal) - Services to the Cargo Use of Operational Area - All traffic (except transhipment) (n64)	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
	Use of Operational Area - All traffic (except transhipment) (n64) Days: 1 - 3 (Free time - included in Standard Senice)	Nature Regulated	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	iee	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
3.3.1 3.3.1.1 3.3.1.2	Use of Operational Area - All traffic (except transhipment) (n64) Days: 1 - 3 (Free time - included in Standard Sentice) Days: 4 - 10 (Fine for total period or fraction period)	Regulated	Per ton	Vessel (Tariff)	Vessel (IGV)	Fr 90.00	ee 16.20	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
3.3.1 3.3.1.1	Use of Operational Area - All traffic (except transhipment) (n64) Days: 1 - 3 (Free time - included in Standard Senice)			Vessel (Tariff)	Vessel (IGV)	Fr	iee	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
3.3.1 3.3.1.1 3.3.1.2 3.3.1.3	Use of Operational Area - All traffic (except transhipment) (in64) Days: 1 - 3 lifees time - induded in Sandrad Sandrae) Days: 4 - 10 lifees total period of racking portion Days: 11 - onwards (Price per day or fraction day)	Requiated Non Regulated	Per ton Per ton/day			Fr 90.00 20.00	16.20 3.60				
3.3.1 3.3.1.1 3.3.1.2 3.3.1.3 Section 3.4	Use of Operational Area. All Intific (except transhipment) (n64)  Days: 1 - 1 Firse time - Incided in Stanting Service)  Days: 4 - 10 (Price for total period or fraction period)  Days: 11 - onwards (Price per day or fraction day)  Special Service of Temporary Depot - Services to the Cargo	Regulated	Per ton	Vessel (Tariff)  Vessel (Tariff)	Vessel (IGV)  Vessel (IGV)	Fr 90.00	ee 16.20	Vessel (Tariff)  Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)  Cargo (Tariff)	Cargo (IGV)
3.3.1 3.3.1.1 3.3.1.2 3.3.1.3 Section 3.4	Use of Operational Area. All traffic (except transhipment) (n64) Days: 1 - 3 (first the included in Standards Standards Days: 4 - 10 (Price for total period or faction period) Days: 4 - 10 (Price for total period or faction period) Days: 11 - oneset of the operation of the oper	Requiated Non Regulated	Per ton Per ton/day Unit of charge			90.00 20.00 Cargo (Tariff)	ee 16.20 3.60 Cargo (IGV)				
3.3.1 3.3.1.1 3.3.1.2 3.3.1.3 Section 3.4 3.4.1 3.4.1.1	Use of Operational Area. All traffic (except transhipment) (n64)  Davs: 1 1 First the - Incided in Stantier Service)  Davs: 4 10 (Price for total period or fraction period)  Days: 1.1 - onwards (Price per day or fraction day)  Special Service of Temporary Depot Integrated Service. All traffic (n65)  Temporary Depot Integrated Service. All traffic (n65)  Re-Ro- carpo in period: bruckless use of operation area until day 10	Regulated Non Regulated Nature	Per ton Per ton/day Unit of charge Per ton			Pr 90.00 20.00 Cargo (Tariff)	16.20 3.60 Cargo (IGV)				
3.3.1 3.3.1.1 3.3.1.2 3.3.1.3 Section 3.4 3.4.1 3.4.1.1 3.4.1.2	Use of Operational Area. All traffic (except transhipment) (n54)  Days: 1 - 3 (refer bein - included in Standard Service)  Days: 4 - 10 (Price for total period or fraction period)  Days: 4 - 10 (Price for total period or fraction period)  Days: 11 - onward (Price per day or faction Gay)  Servical Service of Temporary Dayot - Services to the Cargo  Temporary Dayot Integrated Service - All traffic (s65).  Re-Pic cargo in operate - includes used or operational reas until day 10  Les of cognitional area from day 1 1 owneds (Price per day or fraction day)	Requiated Non Regulated	Per ton Per ton/day Unit of charge	Vessel (Tariff)		90.00 20.00 Cargo (Tariff)	ee 16.20 3.60 Cargo (IGV)				
3.3.1 3.3.1.2 3.3.1.2 3.3.1.3 Section 3.4 3.4.1 3.4.1.1 3.4.1.2 3.4.1.3	Use of Operational Area. All traffic (except transhipment) (n64)  Davs: 1 1 First the - Incided in Stantier Service)  Davs: 4 10 (Price for total period or fraction period)  Days: 1.1 - onwards (Price per day or fraction day)  Special Service of Temporary Depot Integrated Service. All traffic (n65)  Temporary Depot Integrated Service. All traffic (n65)  Re-Ro- carpo in period: bruckless use of operation area until day 10	Regulated Non Regulated Nature Non Regulated	Per ton Per ton/day Unit of charge  Per ton Per ton/day	Vessel (Tariff)	Vessel (IGV)	90.00 20.00 Cargo (Tariff) 120.00	ee 16.20 3.60 Cargo (IGV) 21.60 1.80	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
3.3.1 3.3.1.2 3.3.1.3 Section 3.4 3.4.1 3.4.1.1 3.4.1.2 3.4.1.3 (n65): This is app	Use of Operational Area. All traffic (except transhipment) (n64)  Days: 1 1 (First time - Incided in Stanting Stanting  Days: 1 1) (Price for total period or fraction period)  Days: 1 One-marks (Price per day or fraction day)  Special Service of Temporary Depot Services to the Cargo  Temporary Depot Integrated Service All traffic (n65)  RePo. Cargo in price - incides use of operational rease until day 10  Use of constroal area from day 11 onwards (Price per day or fraction day)	Regulated Non Regulated Nature Non Regulated	Per ton Per ton/day Unit of charge  Per ton Per ton/day	Vessel (Tariff)	Vessel (IGV)	90.00 20.00 Cargo (Tariff) 120.00	ee 16.20 3.60 Cargo (IGV) 21.60 1.80	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
3.3.1 3.3.1.2 3.3.1.2 3.3.1.3 Section 3.4 3.4.1 3.4.1.1 3.4.1.2 3.4.1.3 ((n55): This is applicated of the Stands	Use of Operational Area. All traffic (except transhipment) (n64)  Days: 1 - 10 (Pitcle for Incided in Standard Standard Days: 1 - 10 (Pitcle for Incided in Standard Standard Days: 1 - 10 (Pitcle for Incided in Standard Standard Days: 1 - 10 (Pitcle for India period or fraction period) Days: 1 - 10 (Pitcle for India period or fraction period) Days: 1 - 10 (Pitcle for India period or India period (Pitcle for India perio	Regulated Non Regulated Nature Non Regulated	Per ton Per ton/day Unit of charge  Per ton Per ton/day	Vessel (Tariff)	Vessel (IGV)	90.00 20.00 Cargo (Tariff) 120.00	ee 16.20 3.60 Cargo (IGV) 21.60 1.80	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
3.3.1 3.3.1.2 3.3.1.3 Section 3.4 3.4.1 3.4.1.1 3.4.1.2 3.4.1.3 (n65): This is applicated for the Stands (n65): Prices of s	Use of Operational Area: All traffic (except transhipment) (n64)  Days: 1 - 3 (Price for included in Standard Service)  Days: 4 - 10 (Price for total period or fraction period)  Days: 4 - 10 (Price for total period or fraction period)  Days: 11 - consequent (Price period or fraction period)  Serviced Service of Temporary Depot - Services to the Cargo  Temporary Depot Integrated Service - All traffic (n65)  Re-Re-cargo in operate: includes used or operational area until day 10  Use of construction area from day 11 (owned) Price per day or fraction day)  Other services (n66)  Cardiosis interiors (PAT Temmiss Calso S.A. is appointed as Temporary Depot (code 3014) after cargo has been previously coordinated and accepted by APMTC (Subject to space availability of Service (PAT Temmiss Calso S.A. is appointed as Temporary Depot (code 3014) after cargo has been previously coordinated and accepted by APMTC (Subject to space availability of Service (PAT Temmiss Calso S.A. is appointed as Temporary Depot (code 3014) after cargo has been previously coordinated and accepted by APMTC (Subject to space availability of Service (PAT Temmiss Calso S.A. is appointed as Temporary Depot (code 3014) after cargo has been previously coordinated and accepted by APMTC (Subject to space availability of Service (PAT Temmiss Calso S.A. is appointed as Temporary Depot (code 3014) after cargo has been previously coordinated and accepted by APMTC (Subject to space availability of Service (PAT Temmiss Calso S.A. is appointed as Temporary Depot (code 3014) after cargo has been previously coordinated and accepted by APMTC (Subject to space availability of Service (PAT Temmiss Calso S.A. is appointed as Temporary Depot (code 3014) after cargo has been previously coordinated and accepted by APMTC (Subject to space availability of Service (PAT Temmiss Calso S.A. is appointed as Temporary Depot (code 3014) after cargo has been previously coordinated and accepted by APMTC (Subject to space availability of Service (PAT Temmiss Calso S.A. is appointed	Regulated Non Regulated  Nature  Non Regulated  y). It includes the Standar	Per ton Per toniday  Unit of charge  Per ton Per toniday  Service cargo portion, use of	Vessel (Tariff)  S S of operational area uni	Vessel (IGV) ee section 3.5 til day 10, temporary	Fr 90.00 20.00 20.00 120.00 120.00 10.00 10.00 depot documentation	ee 16.20 3.60 Cargo (IGV) 21.60 1.80 emission (volante) a	Vessel (Tariff)  Individual transmission to C	Vessel (IGV) ustoms. First three o	Cargo (Tariff)	Cargo (IGV)
3.3.1 3.3.1.1 3.3.1.2 3.3.1.3 Section 3.4 3.4.1 3.4.1.1 3.4.1.2 3.4.1.3 (n65): This is appart of the Stands (n65): Prices of s	Use of Operational Area: All traffic (except transhipment) ins6() Days: 1 - 1) Fine the in-included in Standards Standards Days: 1 - 1) Officials for Included in Standards Standards Days: 1 - 1) Officials for India period of Tection period Days: 1 - 1 Officials for India period of Tection period Days: 1 - 1 Consectify Citizens before the Standard Service of Temporary Depot - Services to the Cargo Temporary Depot Integrated Service - All traffic (refs) Re-Ric cargo in general - Publishes used of period and easy officials of the Cargo in general - Publishes are of Cargo in general - Publishes used on Service - All traffic (refs) Re-Ric cargo in general - Publishes used on Service - All traffic (refs) India service - All Temmins Cables - S.A. is agointed as Temporary Depot (note 31(4)) after cargo has been previously coordinated and accepted by APMITC (Subject to space availability of service. Publishes of service and on applicable to large of whole load and accepted by APMITC (Subject to space availability of service. Publishes of service in applicable to large of whole load and accepted by APMITC (Subject to space availability of service in APMITC (Subject to Subject to APMITC (Subject to Subject to APMITC (Subject to Subject to APMITC (Subject to	Regulated Non Regulated Nature Non Regulated	Per ton Per ton/day Unit of charge  Per ton Per ton/day	Vessel (Tariff)	Vessel (IGV)	P0.00 20.00 Cargo (Tariff) 120.00	ee 16.20 3.60 Cargo (IGV) 21.60 1.80	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
3.3.1 3.3.1.1 3.3.1.2 3.3.1.3 Section 3.4 3.4.1 3.4.1.1 3.4.1.2 3.4.1.3 (n65): This is app part of the Stands (n66): Prices of 5 Section 3.5 3.5.1	Use of Operational Area: All traffic (except transhipment) (n64)  Days: 1 - 10 (Pitch for Incided in Standard Standard Standard)  Days: 1 - 10 (Pitch for Incided in Standard Standard Standard)  Days: 1 - 10 (Pitch for India period of Inciden period)  Sexical Service of 1 Emporary Depot Integrated Service: All standard (service of 1 Emporary Depot - Services to the Cargo  Responsive Depot Integrated Service: All standard (service)  ResPot cargoin represent: Excludes use of operational sea until day 10  Level colorational standard service in (service)  Other services (n66)  Listificate Memorar APIT Terminals Calso S.A. is a sponted as Temporary Depot (coto 3014) after cargo has been previously coordinated and accepted by APITT (Subject to space availability of Service. Those of this section are applicable to a tip period device and machiney.  Accepted Calson Service in Service (and Services to Res Ora Cargo (Port Terminal & Temporary Depot) - Services to the Cargo  Handling Services in Storage area	Regulated Non Regulated Nature Non Regulated Non Regulated No Regulated No Regulated No Regulated	Per ton Per toniday Unit of charge Per ton Per toniday  I Service cargo portion, use Unit of charge	Vessel (Tariff)  S S of operational area uni	Vessel (IGV) ee section 3.5 til day 10, temporary	FF 90.00 20.00 20.00 Tariff)  120.00 10.00 depot documentation  Cargo (Tariff)	ee 16.20 3.60 Cargo (IGV) 21.60 1.80 emission (volante) a	Vessel (Tariff)  Individual transmission to C  Vessel (Tariff)	Vessel (IGV) ustoms. First three o	Cargo (Tariff)	Cargo (IGV)
3.3.1 3.3.1.1 3.3.1.2 3.3.1.3 Section 3.4 3.4.1 3.4.1.1 3.4.1.2 (n55): This is app- part of the Stands (n55): Prices of s Section 3.5 3.5.1	Use of Operational Area: All traffic (except transhipment) (n64)  Days: 1 - 10 (Price for Incided in Standard Sarrice)  Days: 1 - 10 (Price for Indiage days of Incident (Price of Incid	Regulated Non Regulated  Nature  Non Regulated  y). It includes the Standar	Per ton Per ton'day Unit of charge Per ton Per ton'day  Service cargo portion, use of Unit of charge Per ton	Vessel (Tariff)  S S of operational area uni	Vessel (IGV) ee section 3.5 til day 10, temporary	Fr 90.00 20.00 20.00 120.00 10.00 depot documentation 11.00 Cargo (Tariff) 11.00	ee 16.20 3.60 Cargo (IGV) 21.60 1.80 emission (votante) a Cargo (IGV) 1.98	Vessel (Tariff)  nd transmission to C  Vessel (Tariff)	Vessel (IGV) ustoms. First three o	Cargo (Tariff)	Cargo (IGV)
3.3.1 3.3.1.1 3.3.1.2 3.3.1.3 Section 3.4 3.4.1 3.4.1.1 3.4.1.2 3.4.1.3 (n65): This is apport of the Stands (n66): Prices of a Section 3.5 3.5.1 3.5.1.1	Use of Operational Area. All traffic (except transhipment) (164)  Days: 1 - 10 (Pitch for Incided in Standard Standard Standard)  Days: 1 - 10 (Pitch for Incided in Standard Standard Standard)  Days: 1 - 10 (Pitch for Intide priority of Incided Incided Standard St	Regulated Non Regulated Nature Non Regulated Non Regulated No Regulated No Regulated No Regulated	Per ton Per toniday Unit of charge Per ton Per toniday  I Service cargo portion, use Unit of charge	Vessel (Tariff)  S S of operational area uni	Vessel (IGV) ee section 3.5 til day 10, temporary	FF 90.00 20.00 20.00 Tariff)  120.00 10.00 depot documentation  Cargo (Tariff)	ee 16.20 3.60 Cargo (IGV) 21.60 1.80 emission (volante) a	Vessel (Tariff)  nd transmission to C  Vessel (Tariff)	Vessel (IGV) ustoms. First three o	Cargo (Tariff)	Cargo (IGV)
3.3.1 3.3.1.1 3.3.1.2 3.3.1.3 Section 3.4 3.4.1 3.4.1.1 3.4.1.2 3.4.1.3 (p65). This is appart of the Stands (p65). Prices of S Section 3.5 3.5.1 3.5.1.1 3.5.1.2 3.5.1.2	Use of Operational Area. All traffic (except transhipment) (n64)  Davis: 1-3 (Price for Incided in Standard Standard Standard)  Davis: 1-1 (Price for total period or fraction period)  Davis: 1-1 (Price for total period or fraction period)  Special Service of Temporary Depot - Services to the Cargo  Temporary Depot Integrated Service: All traffic (n65)  Re-Re-cargo in general - Anches use of operational area until day 10  Use of constrollar area for one 11 of howest Price per day or fraction day)  Other services (n66)  Other services (n66)  Other services (n66)  Temporary Depot Integrated Service: All traffic (n65)  Re-fice cargo in general - Anches use of operational area until day 10  Use of constrollar and one 11 of howest Price per day or fraction day)  Other services (n66)  Temporary Depot (Integrated Services Services (n66))  Temporary Depot (Integrated Services Services Services (n66))  Temporary Depot (Integrated Services Servi	Regulated Non Regulated Nature Non Regulated Non Regulated No Regulated No Regulated No Regulated	Per ton Per ton'day Unit of charge Per ton'day Service cargo portion, use of Unit of charge Per ton Per ton Per ton Per ton	Vessel (Tariff)  S S of operational area uni	Vessel (IGV) ee section 3.5 til day 10, temporary	Fr 90.00 20.00 20.00 120.00 120.00 10.00 10.00 Gepot documentation 11.00 2.50	ee 16.20 3.60  Cargo (IGV)  21.60 1.80  emission (volante) a  Cargo (IGV)  1.98  0.45	Vessel (Tariff)  Ind transmission to C  Vessel (Tariff)	Vessel (IGV) ustoms. First three o	Cargo (Tariff)	Cargo (IGV)
3.3.1 3.3.1.1 3.3.1.2 3.3.1.3 Section 3.4 3.4.1.1 3.4.1.1 3.4.1.2 3.4.1.3 (n65): This is appiant of the Stands (n65): Prices of a (n65): Prices of a 5.5.1.1 3.5.1.1 3.5.1.2 3.5.2.2	Use of Operational Area: All traffic (except transhipment) ins64  Dans: 4 - 10 Price for Incided in Standards Standard  Dans: 4 - 10 Price for Indiagnosis of Incident periods  Dans: 4 - 10 Price for Indiagnosis of Incident periods  Dans: 4 - 10 Price for Indiagnosis of Incident periods  Dans: 4 - 10 Price for Indiagnosis of Incident periods  Dans: 1 - consent of Incident Periods  Dans: 1 - consent of Incident Periods  Dans: 1 - consent Periods  Dans: 1 - consent Periods  Dans: 1 - consent Periods  Dans: 2 - consent Periods  Dans	Regulated Non Regulated Nature Non Regulated Non Regulated No Regulated No Regulated No Regulated	Per ton Per tonday Unit of charge Per ton Per tonday Service cargo portion, use Unit of charge Per ton Per ton Per ton Per ton Per ton Per ton	Vessel (Tariff)  S S of operational area uni	Vessel (IGV) ee section 3.5 til day 10, temporary	FF 90.00 20.00 20.00 120.00 10.00 10.00 depot documentation 11.00 2.50 281.00	ee 16.20 3.60 Cargo (IGV) 21.60 1.80 emission (votante) a Cargo (IGV) 1.98	Vessel (Tariff)  nd transmission to C  Vessel (Tariff)	Vessel (IGV) ustoms. First three o	Cargo (Tariff)	Cargo (IGV)
3.3.1 3.3.1.1 3.3.1.2 3.3.1.3 Section 3.4 3.4.1 3.4.1.1 3.4.1.2 3.4.1.3 (p65). This is appart of the Stands (p65). Prices of S Section 3.5 3.5.1 3.5.1.1 3.5.1.2 3.5.1.2	Use of Operational Area. All traffic (except transhipment) (n64)  Davis: 1-3 (Price for Incided in Standard Standard Standard)  Davis: 1-1 (Price for total period or fraction period)  Davis: 1-1 (Price for total period or fraction period)  Special Service of Temporary Depot - Services to the Cargo  Temporary Depot Integrated Service: All traffic (n65)  Re-Re-cargo in general - Anches use of operational area until day 10  Use of constrollar area for one 11 of howest Price per day or fraction day)  Other services (n66)  Other services (n66)  Other services (n66)  Temporary Depot Integrated Service: All traffic (n65)  Re-fice cargo in general - Anches use of operational area until day 10  Use of constrollar and one 11 of howest Price per day or fraction day)  Other services (n66)  Temporary Depot (Integrated Services Services (n66))  Temporary Depot (Integrated Services Services Services (n66))  Temporary Depot (Integrated Services Servi	Regulated Non Regulated Nature Non Regulated Non Regulated y), It includes the Standar Nature Non Regulated	Per ton Per ton'day Unit of charge Per ton'day Service cargo portion, use of Unit of charge Per ton Per ton Per ton Per ton	Vessel (Tariff)  S S of operational area uni	Vessel (IGV) ee section 3.5 til day 10, temporary	Fr 90.00 20.00 20.00 120.00 120.00 10.00 10.00 depot documentation 11.00 2.50 281.00 1250.00 1	ee 16.20 3.60  Cargo (IGV)  21.60 1.80  emission (votante) a  Cargo (IGV)  1.98 0.45	Vessel (Tariff)  nd transmission to C  Vessel (Tariff)	Vessel (IGV) ustoms. First three o	Cargo (Tariff)	Cargo (IGV)

SECTION 4		SOLID BULK CAF	G0								
Section 4.1	Standard Service	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
4.1.1	Standard Service to the Vessel - Services to the Vessel										
4.1.1.1	Use or rental of berth (Tariff per hour or fraction hour) (n1)	Regulated	Total LoA (m) x Hours	1.55	0% Rate			1.34	Tasa 0%		
4.1.2	Standard Service to Solid Bulk Cargo - Services to the Cargo										
4.1.2.1	Loading or Unloading of Solid Bulk	Regulated	Per ton			8.08	1.45			7.20	1.30
Section 4.2	Special Services - Services to the Vessel	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
4.2.1	Hatches or Hold Lids(n6)										
4.2.1.1	Movement of ISO hatches	Regulated	Per move	0.00	0% Rate						
Section 4.3	Special Services - Services to the Cargo		Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
4.3.1	Remove/place of artificial separation in vessel's hold										
4.3.1.1	Remove of artificial separation in vessel's hold	Regulated	Per operation			1,950.00	351.00				
4.3.2	Use of Operational Area - All traffic										
4.3.2.1	Days: 1 - 5 (Free time - included in Standard Service) (n71)	Regulated				Fre					
4.3.2.2	Days: 6 – 15 (Price for whole period or fraction period)		Per ton			2.90	0.52				
4.3.2.3	Days: 16 – 30 (Price for whole period or fraction period)	Non Regulated	Per ton			4.00	0.72				
4.3.2.4	Days: 31 - onwards (Price per day or fraction day)		Per ton/day			0.40	0.07				
4.3.3	Other services										
4331	Additional weighing (Extra movement is not included) (n72)	Non Regulated	Per truck			21.00	3.78				
	Administrating (Exist intervention is not included) (1172) are given to goods which use absorbing (bowers and sites.										

SECTION 5		LIQUID BULK CAF	RGO								
Section 5.1	Standard Service	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
5.1.1	Standard Service to the Vessel - Services to the Vessel										
5.1.1.1	Use or rental of berth (Tariff per hour or fraction hour) (n1)	Regulated	Total LoA (m) x Hours	1.55	0% Rate			1.34	Tasa 0%		
5.1.2	Standard Service to Liquid Bulk Cargo - Services to the Cargo										
5.1.2.1	Loading or Unloading of Liquid Bulk	Regulated	Per ton			4.61	0.83				
Section 5.2	Special Services - Services to the Cargo	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
5.2.1	Use of Contention Barriers										
5.2.1.1	Use of Contention Barriers (Tariff per day or fraction day) (n73)	Regulated	Per day			1,600.00	288.00			1,500.00	270.00
	icable for the use of contention barriers during loading and unloading operations of hazardous substances and/or goods such as hydrocarbons and other liquid bulk. Service includes placing	and unplacing contention	barriers as well as cleaning a	and drying of contents	on barriers. Tariff of t	his service during load	ling/unloading opera	tions of hazardous li	quid bulk shall be ap	plicable to the cargo or	ansignee, unless

SECTION 6

6.1.1	Standard Service to the Vessel - Services to the Vessel									
6.1.1.1	Use or rental of berth (Tariff per hour or fraction hour) (n1)	Regulated	Total LoA (m) x Hours	1.55	0.28					
6.1.2	Standard Service per Passenger									
	Standard Service per Passenger (n74)	Regulated	Per Passenger			18.15	3.27			
(n74) It applies to	every passenger embarked or disembarked, if the same passenger disembarks and disembarks (on the same or different vessel) the tariff is applied each time. Children under 12 years old an	nd crew members are not	subject to charge.							
SECTION 7	OTHER COL	MMON SERVICES TO	THE VEGGET (#76)							
SECTION	Office Con	MINION SERVICES TO	ITIE VESSEE (IIIS)							
Section 7.1	Special Services - Services to the Vessel	Nature	Unit of charge	Vessel (Tariff)	Vessel /IGV)	Carno (Tariff)	Carno (IGV)	Vessel (Tariff) Vessel (IGV)	Carno (Tariff) Carl	erno (IGV)

SECTION 7	OTHER COL	MMON SERVICES TO	THE VESSEL (n75)								
Section 7.1	Special Services - Services to the Vessel	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
7.1.1	Use of Contention Barriers										
7.1.1.1	Use of Contention Barriers (Tariff per hour or fraction hour) (n76)	Regulated	Por día	1,600.00	288.00			1,500.00	270.00		
7.1.2	Hold Cleaning (n77)										
7.1.2.1	Vessel Holds cleaning (price per gang-shift or fraction) (n78)		Per gang/Shift	510.00	91.80						
7.1.2.2	Unloading of non-hazardous solid waste (price per M3 or fraction) (n79)	Non Regulated	Per M3	255.00	45.90						
	Unloading of hazardous solid waste (price per M3 or fraction) (n80)		Per M3	400.00	72.00						
	Sludge Management										
	Oily Waste Management (n81)		Per truck	1,326.00	238.68						
	Unloading of solid waste (price per M3 or fraction) (n82)	Non Regulated	Per M3	357.00	64.26						
	Management of non-Hazardous solid waste - non recoverables (price per M3 o fracción) (n83)	Ivoi i vogulateu	Per M4	357.00	64.26						
7.1.3.4	Management of Hazardous solid waste (price per M3 or fraction) (n84)		Per M3	500.00	90.00						
7.1.4	Other services										
	Water supply (n85)	Non Regulated	Per ton	10.20	1.84						
	Phone service	Non Regulated	Per activity	Per q	uote						
(n75) Services app	licable to all type of vessels.										

(5)) Survives applicable to all great developed.

(6)) This as explained to add great of the result of controlled to be applicable to the properties of the result of controlled to be applicable to the properties of a controlled to be applicable to discharge one of compatible goods or same vessed call, in other words, to discharge great or applicable to discharge great g

(ASS). This service melters to discharging or collecting from vessels non-econversable waste classified as non-designeous. Service includes temporary store in the terminal facility if necessary, as well as final waste disposal. Price of service applies to those cases where APM Terminals Callato S.A. does not carry out vessel hold cleaning service juscino 7.1.21, Minimum invoice shall be USD 357.

(n81). This service refers to discharging or collecting from vessels hazarchus solid wastage, Service includes temporary store in the terminal facility if necessary, as well as final waste disposal. Price of service applies to those cases where APM Terminals Callso S.A. does not carry out vessel hold cleaning service (section 7.12). Minimum morace shall be 155 500.

SECTION 8	COMMERCIAL POLICIES (r/86)										
Section 8.1	Cancellation of loading	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
8.1.1	Cancellation of container loading and withdraw of container (n87)										
8.1.1.1	20' Container	Non Regulated	Per container			217.00	39.06				
8.1.1.2	40' Container	Non Regulated	Per container			272.00	48.96				
8.1.2	Cancellation of break bulk and ro-ro cargo loading (n88)										
(n86): For further	setals on the scope and conditions of applicable discounts, offers and promotions, see Charter V of Tariffs Regulation and Commercial Policy of APM Terminals Callao S.A. published on well	n85) For further details on the scope and conditions of applicable discounts, office and promotions, see Charler V of Tariffs Regulation and Commercial Policy of APM Terminals Callao S.A. published on weballe: <a href="https://www.apmterminaiscallao.com/pe">https://www.apmterminaiscallao.com/pe</a>									

(167). This is sociable whenever the user (consistence recreashables or otherwise a container of section 1.3 (use of consisting stays and container of stays and container will office a container of section 1.3 (use of consisting stays and container of section 1.3 (use of container stays and container of section 1.3 (use of container stays and container of section 1.3 (use of container stays and container stays 

ro-ro cargo stay longer than the free time granted by the standard service to these kinds of cargo.

Section 8.2	Berthing window reservation	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
8.2.1	Contracting of berth window reservation for vessels										
8211	Contracting of guaranteed berth window reservation for regular traffic vessels with ro-m camp	Non Requisted	by vessel landfall	10 000 00	1 800 00						

SECTION 9	SURCHARGES (n89)							
Section 9.1	General surcharges	Unit of charge	USD (\$)	IGV	Total			
9.1.1	Certificates and receipts							
9.1.1.1	New receipt printing	Per receipt	10.00	1.80	11.80			
9.1.1.2	Issue of weight certificate	Per document	5.00	0.90	5.90			
9.1.1.3	Issue of copy of weighing ticket and/or tally note	Per document	5.00	0.90	5.90			
9.1.1.4	Letter of amendment and/or sending new details to SUNAT	Per document or new transmission	20.00	3.60	23.60			
9.1.1.5	New collection settlement (n90)	Per document	8.00	1.44	9.44			
9.1.2	Returned checks and default (n91)							
(n90): This applies	information and acope, refer to Charter V, section 5.6 of the Tariffs Regulation and Commercial Policy of APM Terminals Callao S.A., published on website: www.apmierminalscallao.com.pe when a collection settlement is generated more than once by the user.							

(n91): Surcharge not subject to IGV. See section 5.6 f the Tarriffs Regulation and Commercial Policy of APM Terminals Callao S.A.

Section 9.2	Surcharges applicable to the Vessel	Unit of charge	USD (\$)		Total
9.2.1	Surcharges for non-use or non-release				
9.2.1.1	Offset for not used crew (n92)	Per crew/hour	162.00	29.16	191.16
9.2.1.2	No release of berth	Per hour	600.00	Not subject to IGV	600.00
9.2.2	Liquid Spill Management Surcharges				
9.2.2.1	Liquid spill management (n93)	Per case	Subj	ect to price quote	

orth Terminal due to user's responsibility. It includes removing and/or collecting and final disposal of the substances, in accordance with current regulations. This surcharge will apply only when APMT carries out

Section 9.3	Surcharges applicable to the Cargo	Unit of charge	USD (\$)		Total
9.3.1	Change of status				
9.3.1.1	Change of container status	Per container	48.90	8.80	57.70
9.3.1.2	Change of non-containerized cargo status	Per B/L	108.60	19.55	128.15
9.3.2	Late arrival				
9.3.2.1	Late arrival of containers	Per container	150.00	27.00	177.00
9.3.2.2	Late arrival of break bulk cargo	Per ton	2.40	0.43	2.83
9.3.2.3	Late arrival of Ro-ro cargo	Per unit	150.00	27.00	177.00
9.3.2.4	Late arrival of solid bulk cargo	Per unit	1.50	0.27	1.77
9.3.2.5	Late arrival of liquid bulk cargo (n94)	Per unit	1.00	0.18	1.18
9.3.3	Late submission				
9.3.3.1	Late submission of shipping documents of containers (n95)	Per DAM	80.00	14.40	94.40
9.3.4	Missing, cancelling or rescheduling appointments (n96)				
9.3.4.1	Missing an appointment (n97)	Per appointment	15.00	2.70	17.70
2.3.4.2	Cancelling or rescheduling an appointment (n98)	Per appointment	10.00	1.80	11.80
9.3.5	Rescheduling services				
9.3.5.1	Rescheduling movement for containers (n99)	Per container	100.00	18.00	118.00
9.3.6	Other surcharges to the Cargo				
9.3.6.1	Bulk cargo re-delivery to truck (n100)	Per truck	225.00	40.50	265.50
9.3.6.2	Liquid spill management (n93)	Per case	Por	cotización	

(46). This applies only to tripping through tenter trucks.

(465). This applies and the product of continues that are taken into the Temporary Depol (5014) out of the cut of set by APM TERMANLS CALLAO S.A.

(465). Appointment will last one (10) tour.

(467). Manage an appointment means a user of set one of set of the appointment for the set of the appointment for the set of the appointment within sight and one hour before the appointment.

(465). The conceilation surchange applies when the user cances the appointment within sight and one hour before the appointment. No conceilation is unchange applies when the user cances the appointment within sight and one hour before the start of the appointment. No conceilation is unchange applies when the cutter or its representative in the section of the appointment of the section of the appointment of the section of the appointment of

DISCOUNTS, OFFERS AND PROMOTIONS
Discounts, desired promotions are commonly accepted practices that answer to compensatory circumstances which shall be determined by APM TERMINAS CALLAD S.A. in compliance with the principle of non-discrimination, equality, neutrality, prohibition of price transference and the election. It is be specified. The EMPAINAS CALLAD S.A. in compliance with the principle of non-discrimination, equality, neutrality, prohibition of price transference and the election. It is be specified from the election of the principle of non-discrimination, equality, neutrality, prohibition of price transference and the election. It is be specified from the election of the principle of non-discrimination, equality, neutrality, prohibition of price transference and the election.

Discounts
Discounts shall be effective during the period established by APM TERMINALS CALLAC S.A. User shall consider the possibility that discounts may not be ren

Promotions and/or temporary offers

APIN TERMINALS CALLAD S.A. may emporarily provide, as part of its commercial policy, offers and/or promotions, as long as permanent equivalent conditions remain.

For cases fuscions in 65th of 51.2 of 1srff in Regulation and Commercial Policy, APIN TERMINALS CALLAD S.A. shall publish on its website the effective term for temporary discounts, promotions and/or offers with details for the adnowledgement of users.

APIN TERMINALS CALLAD S.A. shall also publish termination date or modification of discounts, promotion and/or offers on its website on time.

CRITERIA POR THE APPLICATION OF DISCOUNTS, OFFERS AND PROMOTIONS
APM TERMINALS CALLAD S.A. shall provide discounts, offers and/or promotions to fees of current Tariffs and/or Prices applicable to the Vessel and/or the Cargo considering the following criteria:

APA TESTANIALS CALLO S. A sala provide discounts, offers and/or pornotions to fees of current Tariffs and/or Prices applicable to the Vessel and/or the Carpo considering the following criteria:
Delated competitiveness
Provinction of these services that could not have been provided without the application of discounts and/or pornotions.
Generation of more productivity,
The case of Services provided by the Application of these.
The case of Services provided by the Application of the Ap

investation (ISU Hatches) sage of containers (Hold or vis quay) guntoading and delivery no ISO/OOG (Additional) - container 20 and 40 foot tall movement in the Multipurpose North Terminal

- o, pection and monitoring g and disassembling of clip system on reefer Gens

cial Services shall be provided considering principles set forth in the Tariff's Regulation and Commercial Policy, and the Concession Contract, so APM TERMINALS CALLAD S.A. and the User may not bridge rules and practices that may after competition of Terminal Set For such reason, above detailed port services shall be provided as per regular and reasonable operations that Foreign Trade demands.

Description details of each service are included in Chapter VII of Services Summary of the Tariff's Regulation and Commercial Policy

RESERVE OF GUARANTEED BETHING WINDOW

Space may be reserved for Visiaal's docking provided that this has been previously amanged between AFM TERMINALS CALLAD S.A. and the shipping lines as per terms and conditions set forth in the Operations Regulations and procedures from the Berthing Window Reservation for Regular Service Vess

APPLICATION OF TAX ON GENERAL SALES (60)
The roles and prices of standard and special services based at zero percent (0%) cits, which by version 4.0 of Tariff Scheme were fisted as exempt from VAT, refer to those services according to the VAT Law considered as exports of services. Rolle of 0% will apply as long as the User is considered under the category referenced caregory conference (10%) cits and a services according to the VAT Law considered as exports of services. Rolle of 0% will apply as long as the User is considered under the category referenced caregory conference (10%) cits and a service (10%) cits and

PAYMENT CONDITIONS

Currency of Payment: Tariff and prices in American Dolans (USS) may be cancelled in Nuevors Soles (SL) according to the average safe exchange rate of the financial system indicated by Superintendence of Bank, Insurance and AFP (SSS) at the date of p

Form of Payment: Payments may be through bank account transference, deposit or by check in the same bank that for this ourpose will be assigned by APM TERMINALS CALLAD S.A.

Form of Pyrmeth Pyrmeth may be though bank account transference, deposts or cy cross. In the same uses a law or purpose.

3) Classified Cerrices: Import Setter the carge jackbed up from the Terminal or the Subject of Setter Se

### OFFERS AND PRICE REDUCTIONS

## (1) SPECIAL OFFER FOR DRY CARGO CONTAINERS IN APM TERMINALS CALLAO'S TEMPORARY DEPOT

Dear customers and uses.

APAI Terminoles Cliate (APIIT) has set special differs for the containers processed as Temporary Dippol (code 3014) that meet the requirements below, according to the type of operation. Offers include the extension of days free of use of operational area and/or discounts on the price of the integrated temporary deposit service. Each promotion is independent (not cumulative), for example, if the relevance of the price of the integrated temporary deposit service. Each promotion is independent (not cumulative), for example, if the relevance of the price of the integrated temporary deposit service. Each promotion is independent (not cumulative), for example, if the relevance of the price of the integrated temporary deposit service. Each promotion is independent (not cumulative), for example, if the relevance of the price of the integrated temporary deposit service. Each promotion is independent (not cumulative), for example, if the relevance of the integrated temporary deposit service. Each promotion is independent (not cumulative), for example, if the relevance of the integrated temporary deposit service. Each promotion is independent (not cumulative), for example, if the relevance of the integrated temporary deposit service. Each promotion is independent (not cumulative), for example, if the relevance of the integrated temporary deposit service. Each promotion is independent (not cumulative), for example, if the relevance of the integrated temporary deposit service. Each promotion is integrated to the integrated temporary deposit service. The relevance of the integrated temporary deposit service of the integrated temporary deposit service. The relevance of the integrated temporary deposit service of the integrated temporary deposit service of the integrated temporary deposit service. The relevance of the integrated temporary deposit service of the integrated

### 1.1 Discount on Integrated Service Price and extension of free days - dry cargo import

Tagge group: Incorporters of dry cargo containers (consignees or its representatives—outsons brok - Segment: Importers who address dry cargo to APAITC's Temporary Depot.

- Temporary offer: Importers of dry cargo (consignees or its representatives—outsons brokes, log import Temporary Depot. sections 1.4.1; and 1.2.4 of the Tailffull; but below:

(i) 2.200 to 299 containers = 10 extra fee days (resulting in a bala of 20 fee days) -customs brokers, logistics operators, etc.) who guarantee minimum amounts equal to or greater than 200, 300, 800, 1.6 thousands and 3 thousand containers through the APMITC's Temporary Depot will be granted discounts and extensions of the operating area (taking into ac

(ii) 2 300 to 799 containers = USD 40 discount from the price set out in sections 1.4.1.1 and 1.4.1.2 of the Tariff List for import containers and 10 extra free days to those already granted in subsections 1.4.1.1 and 1.4.1.2 of the Tariff (resching a total of 20 free days)

(i) 2 Alon to 3 productions = 105 00 discount from the price set out in sections 1.4.1.1 and 1.4.1.2 of the Tailff List for import containers and 2 or share days to those already granter in subsections 1.4.1.1 and 1.4.1.2 of the Tailff List for import containers and 2 or share because you will be the selection of the selection 1.4.1.1 and 1.4.1.2 of the Tailff List for import containers and 2 or share because you will be subsections 1.4.1.1 and 1.4.1.2 of the Tailff List for import containers and 20 share because you will be subsections 1.4.1.1 and 1.4.1.2 of the Tailff List for import containers and 20 share because you will be subsections 1.4.1.1 and 1.4.1.2 of the Tailff List for import containers and 20 share free days to those already granted in subsections 1.4.1.1 and 1.4.1.2 of the Tailff List for import containers and 20 share free days to those already granted in subsections 1.4.1.1 and 1.4.1.2 of the Tailff List for import containers and 20 share free days to those already granted in subsections 1.4.1.1 and 1.4.1.2 of the Tailff List for import containers and 20 share free days to those already granted in subsections 1.4.1.1 and 1.4.1.2 of the Tailff List for import containers and 20 share free days to those already granted in subsections 1.4.1.1 and 1.4.1.2 of the Tailff List for import containers and 20 share free days to those already granted in subsections 1.4.1.1 and 1.4.1.2 of the Tailff List for import containers and 20 share free days to those already granted in subsections 1.4.1.1 and 1.4.1.2 of the Tailff List for import containers and 20 share free days to those already granted in subsections 1.4.1.1 and 1.4.1.2 of the Tailff List for import containers and 20 share free days to those already granted in subsections 1.4.1.1 and 1.4.1.2 of the Tailff List for import containers and 20 share free days to those already granted in subsections 1.4.1.1 and 1.4.1.2 of the Tailff List for import containers and 20 share free days to those already granted in subsections 1.4.1.1 and 1.4.1.2 of the Tailff

Effective: Importers may apply or request this promotion until June 30, 2023.

The minimum volumes of the offer will be accounted for smouthy. That is, in the 12 months following the date of the agreement (between APMIC and the user) to start accounting to the volumes (for example and as a hypothetical case; if the user's request for offer is accepted and it is agreed to start accounting on October 1, 2022, then the accounting will be made until September 30, 2023). For this purpose, users must submit to the commercial area of APMIC the projections of containers to be imported in the annual period indicated above, in order to determine the availability of space at the terminal.

In the near that the near think to come think to comply with entering the committed volume of containers into the DT 3014 within 12 months, an additional period of six (8) months may be greated to comply with the committed volume. This additional period is long as it has been verified that, in the first 12-month accounting period, the user has entered into the DT 3014 10015 of the containers

If, after the event of the bester (12) months of accounting (or the 15 months in the cases that apply the extension of ferm), it is welfied that the minimum volumes required for these offers have not been reached, APATC will apply a penalty (the most broundable) in the user. The openses have the considered are as Stote to considered are as Stote to considered are as Stote to considerable as Stote to account the only one penalty will apply and it will be the one that is most broundable to the user. The openses for exclusive the reached accounting (or the 15 months in the cases that apply the extension of term), it is welfer that the minimum volumes required for these offers have not been reached. APATC will apply a penalty (the most broundable) to the user. The openses for a factor that the contraction of the stote of the stot

Part IV all charge, as appropriate, the prices established in subsection 1.4.1 of the Tariff List in force at the time of verification (which refers to a package of services and the use of the operating sees in temporary Depth 2014 for day import containers). That is, in the case of discounts, the prices of subsection 1.4.1.2 of the Tariff List (which refers to be the service appearance to the prices of the operating sees will be a brigged and will be a brig

The contract of the contract o

age to easy just notes as the sums or extraction.

If on the part of the part

Part No. 2 in the case of discourts, the user will pay the amount of the discourt on the number of containers missing to rescrib the larget committed to APATIC. In the case of the additional fee days that have been granted, the penalty applies to the containers that have floally entered the DT 3014 and its application is the same as that described in penalty; that is, in case of no.

The fee days that coveragond to the promotion range; (x) in which the volume floally entered the DT 3014 and its application is the same as that described in penalty; that is, in case of no.

The fee days that coveragond to the promotion range; (x) in which the volume floally entered the DT 3014 and its application is the same as that described in penalty; that is, in case of no.

The fee days that coveragond to the promotion range; (x) in which the volume floally entered the DT 3014 and its application is the same as that described in penalty; that is, in case of no.

The fee days that one the promotion range; (x) in which the volume floally entered the DT 3014 and its application is the same as that described in penalty; that is, in case of no.

The fee days that of the penalty applies to the containers that have floally entered the DT 3014 and its application is the same as that described in penalty; that is, in case of no.

The fee days that of the penalty applies to the containers that have floally entered the DT 3014 and its application is the same as that described in penalty; that is, in case of no.

The fee days that the penalty applies to the containers that have floally entered the DT 3014 and its application is the same as that described in penalty; that is, in case of no.

The case of discourts will penalty applies the containers that have floally entered the DT 3014 and its application is the same as that described in penalty applies the days applied to the penalty applies the containers that the penalty applies the containers that the days applies that the penalty applies that the penalty applies that the penalty app

For example, The user accessed the last damped if the personation from the Company of the Compan

Another example of penalty 2, if the user accessed the burth range of the promotion (from 1,500 to 2,999 containers) and pointed to enter 2,000 containers - for which a discount of USD 30 per container and 20 additional free days would apply, which means reaching a blad of 30 days of use of the operational area. Just in the access of the few days greatly which means reaching a blad of 30 days of use of the operational area. Just in the access of the few days greatly which means reaching a blad of 30 days of use of the operational area. Just in the access of the few days greatly and the promotion area. The manufacture of the access of the few days greatly and the promotion area. The manufacture of the access of the few days greatly and the access of the acc

This offer unifies promotions 1.1 and 1.4 described in version 11.1 of the Tariff List without altering in any way the benefit granted to users.

In case of changes in the prices of the subsections 1.4.1.1 o 1.4.12 of the Tariff List, the discounts of USD 40, USD 50 or USD 60 apply to these modified prices in force at the time of invoicing. See invoicing details in the Regulation of Tariffs and Commercial Policy of APM Terminals Callao

- This offer is subject to availability of space determined by APMTC.

## nt on Integrated Service Price and extension of free days – dry cargo export

- Target group: General users (exporters, forwarders or its representatives –customs brokers, logistics operators, etc. and other users) of containers with dry export cargo.

Segment: Exporters who address dry cargo to APMTC's Temporary Depot.

Tempopary offer Equates (forwards or it is representatives autumn protests, bogistics operators, etc.) who guarantee minimum amounts of more than 300, 700, 800, and 3000 containers using the APMTC's Temporary Depot will be granted discounts and end of 1.4.2.2 of the Tariff Liaf, according to the basic below:

(i) = 700 hasts 799 contenedores = USD 20 discount from the price set out in sections 1.4.2.2 of the Tariff Liaf for export containers and 3 exits free days (resulting in a basic of 10 fee days)

2 of the latter Ladj, accorately to the state below.

(a) 2 MD 069 Contracts — 3 exists free deply resulting in a total of 10 free days).

(b) 2 MD 069 Contracts — 3 exists free days (resulting in a total of 10 free days).

(c) 2 MD 060 2 MD 060

Effective: Exporters may apply or request this promotion until June 30, 2023.

The initiation volumes of the offer will be accounted for annually. In this is, in the 12 months distinctioning the date of the agreement (between ARTIT can't be instanced to example, and is a hypothetical case, if the user's request for offer is accepted and it is agreed to dark accounting on October 1, 2022, then the accou

in the event that the user fails to comply with entering the committed volume of containers into the DT 3014 within 12 months, an additional period of six (05) months may be granted to comply with the committed volume. This additional period will be granted as long as it has been verified that, in the first 12-month accounting period, the user has entered into the DT (3014) 100% of the committed volume. This additional period will be granted as long as it has been verified that, in the first 12-month accounting period, the user has entered into the DT (3014) 100% of the committed volume. This additional period will be granted as long as it has been verified that, in the first 12-month accounting period, the user has entered into the DT (3014) 100% of the committed volume. This additional period will be granted as long as it has been verified that, in the first 12-month accounting period, the user has entered into the DT (3014) 100% of the committed volume. This additional period will be granted as long as it has been verified that, in the first 12-month accounting period, the user has entered into the DT (3014) 100% of the committed volume. This additional period will be granted as long as it has been verified that, in the first 12-month accounting period, the user has entered into the DT (3014) 100% of the committed volume. This additional period will be granted as long as it has been verified that the user has a committed volume. This additional period will be granted as long as it has been verified that the user has a committed volume. This additional period will be granted as long as it has been verified that, in the first 12-month accounting period in the transfer of the DT (3014) 100% of the committed volume. This additional period will be granted as long as a committed volume of the DT (3014) 100% of the DT (3

f, after the not of the bester (T) months of a doccursing (or the 15 months in the cases that apply the extension of lenn), it is welfard that the income that is most descurable to the contract of the bester (T) months of the contract of account to one that is most descurable to the contract of the length of the contract of the cont

The parameters to the considered are as follows:

| Parameter | Pa

Another example of penalty 1, if the user accessed the burth range of the promotion (equal to or more than 3 thousand containers) and promised to enter 4,000 containers - for which a discount of USD 20 per container and 13 additional free days would apply, which means reaching a bid of 20 days of use of the operational great, and in the case of the decount, the user would have to pay for the 200 containers entered into 13 164 fee prices of subsections 14.2 is and 14.2 and 14.

Pauls 2 in the case of discounts, the user will pay the amount of the discount on the number of containers missing to reach the larget committed to APVITC. In the case of the additional fee days that have been gentled, the penalty applies to the containers that have finally entered the UT 3014 and its application is the same as that described in penalty 1; that is, no case of non-committee the penalty applies to the containers that have finally entered the UT 3014 and its application is the same as that described in penalty 1; that is, no case of non-committee the penalty applies to the containers that have finally entered the UT 3014 and its application is the same as that described in penalty 1; that is, no case of non-committee that the penalty applies to the containers that the original penalty applies to the containers that there finally entered the UT 3014 and its application is the same as that described in penalty 1; that is, no case of non-committee that the penalty applies to the containers that there finally entered the UT 3014 and its application is the same as that described in penalty 1; that is, no case of non-committee that the penalty applies to the penalty applies to the containers that there finally entered the UT 3014 and its application is the same as that described in the penalty 1; that is, no case of non-committee that the penalty applies to the containers that the penalty applies to the containers that the finally entered the UT 3014 and its application is the same as that described in the penalty applies to the penalty applies to the penalty applies to the containers that the penalty applies to the penalty

For example, If the use accessed the find range of the pomotion (in miles 10 15 50 containers), and possible of the 50 containers and the second of the 15 containers and the 15 containers are the second of the 15 containers and the 15 containers are the

rest or restrictions.

The contraction of the contr

This offer unifies promotions 1.2 and 1.5 described in version 11.1 of the Tsriff List without altering in any way the benefit granted to users.
In case of changes in the prices of the subsections 1.4.2.1 or 1.4.2.2 of the Tsriff List, the discounts of USD 20 apply to these modified prices in force at the time of invoicing. See invoicing details in the Regulation of Tsriffs and Commercial Policy of APM Terminals Callact

## 1.3 Discount on Integrated Service Price and extension of free days - Import/Export of dry cargo

L3 Unscount on managements deriver, increase accumumation on mine quays — import.cuptor or any curgot — Traget group. General laws group frest/prosspace service and acquired that groups of the process of the process

(i) > 300 to 699 containers = 10 additional free days with respect to the free days of use of operational area indicated in sections 1.4.1.1 and 1.4.1.2 of the Tariff List for import containers (entainer security in a total of 20 free days) and 3 additional free days with respect to the free days indicated in sections 1.4.2.1 and 1.4.2.2 of the Tariff List for export containers (excluding in a total of 10 free days). (i) 2 700 to 1799 containers = 1800 40 discount and 10 additional free days with respect to the prices and free days indicated in sections 1.4.2.1 and 1.4.1.2 of the Tairff List for import containers (resulting in a total of 00 free days) and USD 20 descount and 3 additional free days with respect to the prices and days indicated in sections 1.4.2.1 and 1.4.2.2 of the Tairff List for export containers (resulting in a total of 00 free days).

a botal of 10 the days)
a 1580 or officence (asys)
1.580 or officence (iii) ≥ 800 to 1,599 (resulting in a total

lv) ≥ 3.000 conf ers = USD 60 discount and 20 additional free dave with respect to the prices and free dave indicated in sections 1.4.1.1 and 1.4.1.2 of the Tariff List for import containers (resulting in a total of 30 free dave) and USD 20 discount and 13 additional free dave with respect to the prices and dave indicated in sections 1.4.2.1 and 1.4.2.2 of the Tariff List for import containers (resulting in a total of 30 free dave) and USD 20 discount and 13 additional free dave with respect to the prices and dave indicated in sections 1.4.2.1 and 1.4.2.2 of the Tariff List for import containers (resulting in a total of 30 free dave) and USD 20 discount and 13 additional free dave with respect to the prices and dave indicated in sections 1.4.2.1 and 1.4.2.2 of the Tariff List for import containers (resulting in a total of 30 free dave) and USD 20 discount and 13 additional free dave with respect to the prices and dave indicated in sections 1.4.2.1 and 1.4.2.2 of the Tariff List for import containers (resulting in a total of 30 free dave) and USD 20 discount and 13 additional free dave with respect to the prices and dave indicated in sections 1.4.1.1 and 1.4.2.2 of the Tariff List for import containers (resulting in a total of 30 free dave) and USD 20 discount and 13 additional free dave with respect to the prices and dave indicated in sections 1.4.1.1 and 1.4.2.2 of the Tariff List for import containers (resulting in a total of 30 free dave) and USD 20 discount and 13 additional free dave with respect to the prices and dave indicated in sections 1.4.1.1 and 1.4.1.2 of the Tariff List for import containers (resulting in a total of 30 free dave) and 1.4.1.2 of the Tariff List for import containers (resulting in a total of 30 free dave) and 1.4.1.2 of the Tariff List for import containers (resulting in a total of 30 free dave) and 1.4.1.2 of the Tariff List for import containers (resulting in a total of 30 free dave) and 1.4.1.2 of the Tariff List for import containers (resulting in a total of 30 free dave) and 1.4.1.2 of the Tari (resulting in a total of 20 free days)
Users may apply or request this offer until June 30, 2023.

The institution volumes of the offers will be accounted for entangle in the comment for entangle in the comment for entangle in the comment of the under in equest for promotion is accepted and it is agreed that the posting on Orbite 1 , 2022, then the accounting will be made until September of the comment of the comment of the under in the

In the event that the user fails to comply with ethering the committed volume of containers into the OT 3014 within 12 months, an additional period of six (05) months may be granted to comply with the committed volume. This additional period will be granted as long as it has been verified that, in the first 12 month accounting period, the user has entered into the OT (3014) 100% of the containers in the committed volume. This additional period will be granted as long as it has been verified that, in the first 12 month accounting period, the user has entered into the OT (3014) 100% of the containers in the committed volume. This additional period will be granted as long as it has been verified that, in the first 12 months accounting period. The user has entered into the OT (3014) 100% of the containers in the committed volume.

f, ther the word the bester (T) control of socrating for the 15 months in the cases that apply the extension of term), it is welfard that the minimum volumes required for these offers have not been reached, APMTC will apply a penalty (the most broundshie) to the user. Two oppies of penalties will be tablen into account, but only one penalty will apply and it will be the one that is most broundable to the user. Two oppies of penalties will be to account the only one penalty will apply and it will be the one that is most broundable to the user. Two oppies to be contented are as both to content are as as as as as as as a section of the content are as as as as a section of the content are as a section of the content are

Park 1. APRITC will chaige, as appropriate, the prices established in the subsections 1.4.1 and 1.4.2 of the Tariff List in force at the time of verification (which relets to be package of services and the use of operational area in Temporary Depot 2014 for dy import and export containers). That is, in the case of discounts, the prices of subsections 1.4.1 and 1.4.2 of the Tariff List (which refers to the temporary depot service package that includes 10 days here for use of the operation uses [or the case of depots on the depots of the depots

For example, if the user accessed the fourth range of the promotion (from 1,000 to 2,200 containers) and pointed to entire 2 housand containers (11 housand of lexport) therefore, in the case of imports, a decount of USD 50 per container and 20 additional the days, would correspond, which means reaching a total of 15 days of use of the operational area, and in the case of imports, a decount of USD 50 per container and 20 additional the days, which means reaching a total of 15 days of use of the operational area, but in the accessor in the case of imports and 1500 days of use of the operation and with regard to the days of the containers are the accessor in the case of imports and 1500 days of use of the operation and the regard to the days of the containers are the accessor in the case of imports and 1500 days of use of the operation and the containers are the case of imports and 1500 days of use of the operation and the containers are the case of imports and 1500 days of use of the operation are the containers are the case of imports and 1500 days of use of the operation are the containers are the case of imports and 1500 days of use of the operation are the containers are the case of imports and 1500 days of use of the operation are the containers are the containers are the case of imports and 1500 days of use of the operation are the containers are the cont

Pently 2, in the case of discounts, the user will pay the amount of the discount on the number of containers missing to reach the beingst committed to APVITC. In the case of the additional fees days that have been grained, the pentally applies to the containers that have founly entered the CT 3014 and its application is the same as that described in pentally 1, that is, in case of non-compliance, the fee days that coveragend to the promotion range (D) is which the he volume finally entered will be estigated and will be dauged for the caley for the days feel the best possible of successions to the new debut range (D), in case of non-compliance, and the contracting any promotion range, the pices of subsections 1.4.1, 1.4.1 and 1.4.5 will be charged in the same of any possible to the contracting any promotion range, the pices of subsections 1.4.1, 1.4.1 and 1.4.5 will be charged in the same of any possible to the contracting any possible contracting any possible to the contracting and possible to the contracting any possible to the contracting and the contracti

For example, The user accessed the both range of the promotion (the 1,000 to 25% containers) and pointed to be refer 2 housed containers (1 housed of import and 1 housed of export). Here were a container (1500 to 25% containers and 25% does not be refer 20 housed of export and 1 housed of export a

notions 1.3 and 1.6 described in version 11.1 of the Tariff List without altering in any way the benefit granted to users.

In case of changes in the prices of the subsections 1.4.1 or 1.4.2 of the Tariff List, the discounts apply to these modified prices in force at the time of invoicing. See invoicing details in the Regulation of Tariffs and Commercial Policy of APM Terminals Callao

- Target group. General users (importers/consignees and exporters/forwarders or its representatives - austroms brokes, logistics operators, etc. - and other users) of containers with import/export dry cargo. This offer is independent of the offers isled above (1,1 to 1.3, 1) that is, those who are granted this promotion will not be eligible for offers from 1.1 to 1.3. Applies to new

Segment: Users who address dry cargo to APMTC's Temporary Depot.

-Temporary offer: Users who mobilize import and export day cargo containers who guarantee minimum amounts equal or more than 150, 400, 800 and 1,500 containers through the APRITC's Temporary Depot will be granted a discount on the price of the Special Integrated Temporary Depot Service for Import and Export Dry Containers (Sections 1.4.1.1, 1.4.2.1 and 1.4.2.2 of the Tariff Legi and an extension of days fee of user of specially integrated area, according to the bable below. is a days fee of use of operational area, according to the bable below.

(a) a 15th 395 containers - For import containers a decount of USD 300 on the price indicated in sections 1.4.1.1 and 1.4.1.2 of the Tariff List (reaching a basid of 100 the days), while for export containers a decount of USD 300 on the price indicated in sections 1.4.1.1 and 1.4.1.2 of the Tariff List (reaching a basid of 100 the days), while for export containers a decount of USD 300 on the price indicated in sections 1.4.1 and 1.4.1.2 of the Tariff List (reaching a basid of 30 the days), while for export containers a decount of USD 300 on the price indicated in sections 1.4.2.1 and 1.4.2.2 of the Tariff List and 50 the days), while for export containers a decount of USD 300 on the price indicated in sections 1.4.1 and 1.4.2 of the Tariff List (reaching a basid of 30 these days), while for export containers a decount of USD 300 on the price indicated in sections 1.4.2 and the Tariff List and 50 the basid and 50 the basid and 50 these days), while for export containers a decount of USD 300 on the price indicated in sections 1.4.2 and 1.4.2.2 of the Tariff List and 50 the basid and 50 the basid and 50 the basid and 50 the basid and 50 these days), while for export containers a decount of USD 300 on the price indicated in sections 1.4.2 and 1.4.2.2 of the Tariff List and 50 the basid and 50

Requirements: The volumes of the promotions will be accounted for during the period of validity of the promotion, that is, between January 1, 2023 and June 30, 2023. For these offers there is no prior agreement or commitment of movements by the user vis-à-vis APM Terminals Callao.

### (2) PROMOTION FOR REFRIGERATED CARGO CONTAINERS IN APM TERMINALS CALL AO

APM Terminals Callao (APMTC) has arranged offers for containers full with export refrigerated cargo to be served by APM Terminals Callao (APMTC), either as a Port Terminal or through our Temporary Depot (code 3014), and that meet the considerations indicated in

The response containing that adjust class or understand to a law transport entered class for severe from a resultant entered to the temporary depot of APMITC

- Target group. Exportes (formation or its representatives—cubine bridges, logistics, separates, logistic

-Segment Refigerated caps exposed as effecting their caps to be APRITC's Temporary Open (Fig. 1995). The company of the Refigerated caps exposed two in the company of the Refigerated caps exposed two in the company of the Refigerated caps exposed two in the company of the Refigerated caps exposed two in the company of the Refigerated caps exposed two in the company of the Refigerated caps exposed two in the Company of the Refigerated Caps and the Company of the Compa

### 2.2 For export containers that use APMTC as a port terminal and that are direct loading

Exercise: In the in a sea unit increment with the sea of the Exercise of the E

## (3) OFFER FOR REQUESTING SERVICE PACKAGES IN APM TERMINALS CALLAD'S TEMPORARY DEPOT - FIRST MODALITY

APM Terminals Callao (APMTC) has arranged offers for all dry import containers to be serviced through its Temporary Warehouse (code 3014) for which service packages are requested. The promotions refer to discounts (which imply lower prices) and apply by type of package

Package 1: integrated service for 20-foot import day container judasection 1.4.1.1 of the Tairff List), use of area from day 11 to day 20 judasection 1.4.1.3 of the Tairff List) and deconsolidation/consolidation with forfill judasection 1.5.3.4 of the Tairff List). The total price to be paid for this service package amounts to USD 400 per container.

Package 2: integrated service for 20-foot import day container judasection 1.4.1.1 of the Tairff List), use of area from day 11 to day 20 judasection 1.4.1.3 of the Tairff List), and or area from day 11 to day 20 judasection 1.4.1.3 of the Tairff List), The total price to be paid for this service package amounts to USD 505 per container.

Package 3. integrated service for 40-fact import dry container (subsection 1.4.1.2 of the Tariff List), use of area from day 11 to day 20 (subsection 1.4.1.3 of the Tariff List) and deconsolidation/econsolidation with british (subsection 1.5.3.4 of the Tariff List). The total price to be paid for this service package are Package 4. integrated service for 40-fact import dry container (subsection 1.4.1.2 of the Tariff List), use of area from day 11 to day 20 (subsection 1.4.1.3 of the Tariff List) and consolidation/econsolidation of containers with dry cargo (with forkith) including Galle In/Galle O.d. of empty container (subsection 1.4.1.3 of the Tariff List). ts to USD 500 ne

ction 1.5.3.7 of the Tariff List). The total price to be paid for this service package amounts to USD 615 per cor

The new prices of these offers apply to any user and will be valid from July 2, 2022 to June 30, 2023. Likewise, access to this offer is subject to availability of space confirmed by APMTC

## (4) OFFER FOR APPLICATION OF SERVICE PACKAGES TO CONTAINERS IN APM TERMINALS CALLAO - SECOND MODALITY

Dear customers and users

Group of packets 1- 07 2014 integrated service for 20 or 40 floor import day container (jubsections 1.4.1.1 and 1.4.1.2 of the Tatiff List), container deconsolidation in any of its forms (jubsections 1.5.3.4, 1.5.3.5 and 1.5.3.6 of the Tatiff List) and use of area from yor 1.5.3.4 in the part List) and use of a

Group of packads 3 - DT 3914. Integrated service for 20 or 40 floot import dry container (subsections 1.4.1.1 and 1.4.1.2 of the Taifff List) and partial consolidation/deconsolidation of container with dry carpo (subsection 1.5.3.13 of the Taifff List), For packages that are claimed, a global discount of USD 82.1 for a 20-foot container and USD 99.5 for a 40-foot container will be granted.

Group of packets 4 - a port terminal carries proteined and special processors. In Jun 1, and the packet pac

## (5) OFFER FOR UN 3077 CARGO CONTAINERS CLASSIFIED AS IMO 9 IN APM TERMINALS CALLAG'S TEMPORARY DEPOT

APM Terminals Callso (APMTC) has set that export containers with dangerous cargo UN 3077 (UN 3077) - such as zinc code, abietic acid and others - classified as MO 9 class, to be serviced through its Temporary Depot (code 3014), the price of the special service "Treatment of dangerous cargo IMO 9" will not be applied, in other words, a 100% discount will be applied to the price of subsection 1.5.1.3. This promote the said from January 1, 2021 to June 29, 2023.

## (6) SPECIAL OFFER FOR FISHMEAL IN HIGH-CUBE CONTAINERS

Dear customers and users,

so (APMTC) has set that containers with fishmest to be shipped will not be charged with the tariff of 'high-cube container' special service, subsection 1.5.1.7. This offer will be in force from January 1, 2021 to June 30, 2023

## (7) OFFER FOR TRANSSHIPMENT OF CONTAINERS

APM Terminals Callao (APMTC) has set a offer for transshipment containers that are loaded/unloaded by the North Multi-purpose Terminal. This offer, which refers to a discount on the standard service tariff and extension of free days, is governed by the following terms:

FIRST Terminates Calibor (PATICT) has set at a confer for transphagement containners that are isoacidulunded by the Min Multi-purpose Terminal. This offer, which elects to absocut on the standard service bufff and exhension of fee days, is governed by the following terms.

- Tanget group. Uses (bityping lines or in representative) that more transphagement containners equal to or greater than 1,100 seeply seeper containners and to great than 1,100 seeply seeper containners (absection 1.1.2.8 of the Tariff List), it includes for both sizes of containners (20 and 40 feet) the granting of 10 fee days (which represents 8 additional feet begin to the Market purposes) be the 45 the born court registered by the 45 the

## (8) OFFER FOR GASIFICATION OF FULL CONTAINERS

(i) ≥ 200 to 399 semi-annual containers = USO 20 decount from the promotional turiff indicated in section 1 2.13.1 of the Turiff Us( fluid is, a further of USO 460 per container will be changed), 2 days five of use of the operational area in addition to the 72 hours granted as part of the standard service for export containers (up to a total of 5 days five of the operational area as POT Terminal and the excession of the payments for drawing of status in case of occurrence. In this other is governed by the following:

(i) 2 400 to 199 semi-amusal containers = USD 35 decount from the promotional barill indicated in section 1,2 13.1 of the Tariff List (that is, a barilf of USD 445 per container will be changed), 2 days free of use of the operational area in addition to the 72 hours granted as part of the standard senince for export containers (up to a brief of 5 days free of the use of the operational area as Port Term and the excursation of the payment for change of status in case of cocurrence.

nal fairli indicated in section 1.2.13.1 of the Tariff List (finkt is, a tariff of USD 420 per container will be charged), 2 days free of use of the operational area in addition to the 72 hours granted as part of the standard service for export containers (up to a total of 5 days free of the use of the operational area as Port Terminal) and the (ii) ≥ 800 semi-annual containers = USD 60 discount from the pro exoneration of the payment for change of status in case of occurrer

- Effective: The offer may be requested from October 06, 2022 to March 30, 2023

The distinuin valueme of the offers will be excursed for on a sensin statistic field in a speed that the accounting on November 1, 2002, then the accounting on

Suppose that the process process must some that commences are as an eventure, the projections of contrainers are as uncommence are as a verture. The properties of the process are a verture in the process are a verture. The process are a verture in the process are a verture in the process and the first and in the process and the process are a verture. The process are a verture in the process are a verture in the process and the process are a verture in the process and the process are a verture in the process and the process are a verture in the process and the process are a verture in the process and the process are a verture in the process and the process are a verture in the process and the process are a verture in the process and the process are a verture in the process and the process are a verture in the process and the process are a verture in the verture in the process and the process are a verture in the process and the process are a verture in the verture in

## (9) SPECIAL OFFER FOR BREAK BULK CARGO - TEMPORARY DEPOSIT OF APM TERMINALS CALLAO

APM Terminals Caliso (APMTC) has set a series of special offers for the special service "Integrated Temporary Depot Service" for break bulk cargo. It should be noted that each of the special offers is independent (it is not cumulative); for example, if the user cho ses promotion 9.4 he would not be able to apply to promotion 8.1, 8.2, 8.3 or 8.5. Special offers are governed by the folio

## 9.1 Extension of free days applicable to break bulk cargo in general and to subsection 2.4.1.1 of the Tariff List - Tarcel crouv: Users (importers/consioness amd Exporters/flowarders or its recresentalities — outstoms brokers, lor

This offer will be subject to space availability confirmed by APMTC

- Temporary offer: Users who guarantee minimum amounts of more than 5,000 and 7,000 tonnes using the APMTC's Temporary Depot will be

(i) ≥ 5,000 to 6,999 tonnes = 5 additional days off (on top of the 10 already granted in subsection 2. 4. 1.1 of the Tariff)
(ii) ≥ 7,000 tonnes = 10 additional days off (on top of the 10 already granted in subsection 2. 4. 1.1 of the Tariff)

Effective: Users may apply or request this promotion from May 21, 2021 until June 30, 2023.

Requeriments: The minimum volume of 5,000 or 7,000 tunnes will be accounted for amounts); that is within have'se (17) months from the date of the agreement (Pealween APMTC and the user) to commence the accounting of volumes (for example and as a hypothetical case; if the user's groundoin request is accepted and it is agreed that the accounting will commence on 1 June 2021, then the accounting of volumes (for example and as a hypothetical case; if the user's groundoin request is accepted and it is agreed that the accounting will commence use a hypothetical case; if the user's groundoin request is accepted and it is agreed that the accounting will commence use a hypothetical case; if the user's groundoin request is accepted and it is agreed that the accounting will commence use a hypothetical case; if the user's groundoin request is accepted and it is agreed that the accounting will commence use a hypothetical case; if the user's groundoin request is accepted and it is agreed that the accounting will commence use a hypothetical case; if the user's groundoin request is accepted and it is agreed that the accounting will commence use a hypothetical case; if the user's groundoin request is accepted and it is agreed that the accounting will commence use a hypothetical case; if the user's groundoin request is accepted and it is agreed that the accounting will commence use a hypothetical case; if the user's groundoin request is accepted and it is agreed that the accounting will cannot be used as a hypothetical case; if the user's groundoin request is accepted and it is agreed that the accounting will cannot be used as a hypothetical case; if the user's groundoin request is accepted and it is agreed that the accounting will cannot be user's groundoin request is accepted and it is agreed that the accounting will cannot be user's accepted and it is agreed to the user's groundoin request is accepted and it is agreed to the user's groundoin request is accepted and it is agreed to the user's groundoin request in acceptance of

9.2 Discount in price and extension of free days applicable to non-hazardous break bulk cargo in general and to subsection 2.4.1.1 of the Tariff List (integrated temporary depot service that includes 10 days of use of operating area)

- Target group: Uses (importers/consignees and Expones/forwardes or its representatives – customs brokes, logistics operators, etc.) of break bulk cargo in general not dangerous that demand the Temporary Depot Integrated Service with 10 days of use of

a price and estension of fine days applicable to non-hazardous break bulk cargo in general and to subsection 2.4.1 of the Tariff List (independed temporary depot service that includes 10 days of use of operating area)

- Taged sport butter (independent-burgeters are inferential proportion-burgeters are inf

The minimum volumes of 10 Phousand, 15 Phousand or 20 Phousand to more will be accounted for over a period of nire (80) months, that is, in the 90 months following the date of the agreement (between APAITC and the user) is start the accounting of the volumes (for example and as a hyporhetical case, if the user's promotion request is accepted and it is agreed to start accounting on June 1, 2022, their the accounting will be made until all analyses? 37, 2023, For this purpose, user must admit be the commended and the commended and the promotion of the user's been only period indicated above, in order to determine the availability of space in the terminal.

The event that the user that is be comply with event and the committed volume of treat but accept into the DT 3014 within 50 months, an additional period of three (03) months may be granted to comply with the committed volume. This additional period will be granted as long as it has been verified that, in the first 05-months accounting period, the user has entered into the DT (0314) 100% of the break but cargo backet and unished by the lichtin Terminal.

1, ther the not of the rise (50 months of exceuding of the 12 months in the cases that upply in extraction of term, it is wided that the minimum varieties of the price established in adjacencies 24.1.1 of the Tailf List (integrated service of temporary storage of break but, capp that includes 10 days of use of operative real) and the integrated services of 12.1 of the Principated described 12.2 of the Principate 12

as junce enabuserou in subsection 2.4 i. 2.0 in it is mail. Let (use or see no cupy) in 2.0 in it is mail. Let (use or see no cupy) in 2.0 in it is mail. Let (use or see no cupy) in 2.0 in it is mail. Let (use or see no cupy) in 2.0 in it is mail. Let (use or see no cupy) in 2.0 in 2.0 in it is mail. Let (use or see no cupy) in 2.0 in 2.0 in it is mail. Let (use or see no cupy) in 2.0 in 2.0

9.3 Discount in price and extension of fee days applicable to high density break bulk cargo that can be stacked and to subsection 2.4.1.1 of the Tariff List (integrated temporary depot service that includes 10 days of use of operating area)

- Target group: Users (importencions/grees area Exportens/brownetses or its representatives—customs broken. Logistic operation, exist, of left hardward the Temporary Object integrated Service with 10 days of use of operating area)

- Temporary offer: Uses who guarantee minimum amounts of equal or more the 10 Brownet tomes using the APITICs Temporary Dept, will be granted a discount of USO 3 per ton on the price established in subsection 2.4.1.1 of the Tariff List, extension of 10 additional fee days of use of the operating area with respect to the fee days granter in subsection 2.4.1.1 of the Tariff List (reacting a total of 20 days of use of the operating area with respect to the fee days granter in subsection 2.4.1.1 of the Tariff List, extension of 10 additional fee days of use of the operating area with respect to the fee days granter in subsection 2.4.1.1 of the Tariff List, extension of 10 additional fee days of use of the operating area with respect to the fee days granter in subsection 2.4.1.1 of the Tariff List, extension of 10 additional fee days of use of the operating area with respect to the fee days granter in subsection 2.4.1.1 of the Tariff List, extension of 10 additional fee days of use of the operating area with respect to the fee days granter in subsection 2.4.1.1 of the Tariff List, extension of 10 additional fee days of use of the operating area with respect to the fee days granter in subsection 2.4.1.1 of the Tariff List, extension of 10 additional fee days of use of the operating area with respect to the fee days granter in subsection 2.4.1.1 of the Tariff List, extension of 10 additional fee days of use of the operating area.

This offer is only applicable to goods that can be stacked on at least three levels. If the merchandise can only be stacked on two levels, the offer does not apply, APMTC will define which goods are stackable up to three levels

The infilingual volume of 10 Phospased formers will be accounted for in an annual period, that is, within 12 months of the date of the ageneratify between APINITO and the user jo test accounting for the minimum volume (for example and as a hypothetical case, if the user's promotion request is accepted and it is agreed that accounting will be made until 30 Clotheter APINITO commencial sense the projection of clones to be mobilized in the yearly period indicated above, in order to determine the availability of space in the lemmand.

If all the few of the 12 months of accounting, it is verified that the minimum volume equivale for this often has not been reacted. APINITO will charge the ist price established in subsection 2.4.1.1 of the Turff List (integrated sensice of temporary storage of break but cargo that includes 10 days of use of operative area) as well as the prices for the user of the operating area from they 11 to 30 that has been used (price of abovections 2.4.1.2 and 2.4.1.3 of the Turff List joint period and accounting will be minimum.

In case, the price of subsections 2.4.1.1 and 2.4.1.3 of the Tariff List is modified, the discounts of USD 9 per ton or USD 2.70 per ton iday apply to the modified price and in force at the time of invoicing. See invoicing details in the in the Regulation of Tariffs and Commercial Policy of APM Te

unt applicable to non-hazardous sacks or big bags and to subsection 2.4.1.4 of the Tariff List (integrated temporary depot service that includes 30 days of use of operating area]

- Target group. Uses (importes/consignees amil Exportes/brandfers or its representatives—cultums broken. logistic operation, etc.) of break but cargo (sacks or big bags) and dangerous that demand the Temporary Depot Integrated Service with 30 days of use of operative area (subsection 2.4.1.4 of the Tariff List)

- Segment: Uses of break but cargo that load or united sacks or big bags with non-dangerous goods in large volumes.

- Temporary Operation (but see to load or united sacks or big bags (break but cargo) with guarantee minimum amounts through the Temporary Depot (3014), equal or higher than 10 thousand or 45 thousands tonnes, will be granted a discount on the price of the temporary depot integrated service that includes 30 days of use of the operational area (subsection 2.4.1.4 of the Tariff List), according to the temporary depot integrated service that includes 30 days of use of the operational area (subsection 2.4.1.4 of the Tariff List), according to the temporary depot integrated service that includes 30 days of use of the operational area (subsection 2.4.1.4 of the Tariff List), according to the temporary depot integrated service that includes 30 days of use of the operational area (subsection 2.4.1.4 of the Tariff List), according to the temporary depot integrated service that includes 30 days of use of the operational area (subsection 2.4.1.4 of the Tariff List).

(i) ≥ 10,000 to 39,999 tonnes = Discount of USD 40 per tonne on the price of subsection 2.4.1.4 of the Tariff List
(ii) ≥ 40,000 tonnes = Discount of USD 47 per tonne on the price of subsection 2.4.1.4 of the Tariff List

Effective: In the case of the special offer i) the users may apply or request this offer from February 17, 2022 until June 30, 2023. In the case of the special offer ii) the users may apply or request this promotion from May 04, 2022 until June 30, 2023.

The estimate value of 10 because (figure qual just be excounted for a see as even as a large and product of a see as of a position of 10 because of time see and a second for a see as of a position of 20 because of time see as of a position of 20 because of time see as of a position of 20 because of time see as of a position of 20 because of time see as of a position of 20 because of a position of 20 because of time see as of a position of 20 because of time see as of a position of 20 because of time see as of a position of 20 because of 10 because of 20 be

## 9.5 Price discount applicable to sacks or big bags and to subsection 2.4.1.4 of the Tariff List (integrated temporary depot service that includes 30 days of use of operating area) - Target group: Users (importers/consignees and Exporters/forwarders or its representatives – customs broken, logistics operation, etc.) of break but cargo (sacks or big bags) that demand the Temporary Construction of the Con

- Target group: Users (important/consignees and Exportant/breadless or its representatives—customs broken, logistic operation, etc.) of break bulk usage places or big begin but demands of the Temporary Depot Registed Service with 30 days of use of operative area (publication 2.4.1.4 of the Tariff List)
- Segment Users of the bulk carpy the fact of users a bulk carpy the fact of public bulk carpy and the fact of users a bulk carpy the fact of the public bulk carpy and the fact of the f

(i) ≥ 10,000 tonnes = Discount of USD 50 per tonne on the price of subsection 2.4.1.4 of the Tariff List Effective: This offer will be in force from October 06, 2022 until January 05, 2023.

The minima volume of 10 housand will be accountedor in a quality hask, that is, during the period of visibly of the edity (Chiber 65, 2022 and Jaury 95, 2023), For this quyees, seen must uslent to be commercial seen of APPLIC the projection of three to be minimal volume engineer. APPLIC all charges the established in section 2.4 L.4 of the first facility of the edity of the edity

In case the price of subsection 2.4.1.4 of the Tariff List is modified, the discount of USD 50 per ton apply to the modified price and in force at the time of invoicing. See invoicing details in the in the Regulation of Tariffs and Commercial Policy of APM Terminals Callao.

# 9.6 Price discount applicable to break bulk cargo not dangerous and to subsection 2.4.1.1 of the Tariff List (integrated temporary depot service that includes 10 days of use of operating area) - Target group: Users (importers)consigness and Exporters/brearies or its representatives – customs brokes, logistics operation, etc.) of general break bulk cargo not dangerous that demand the Temporary Depot (DT 3014)

Temporary offer: Users who guarantee minimum amounts of more than 2.5 thousands, 4 thousands and 5 thousands ton of the Tariff List), according to the table below.

(i) a 2,000 to \$90 trens = 10 trens or \$1000 or

The volumes of the promotions will be counted during the ceriod of validity of the promotion; that is, it will be counted between January 1, 2023 and March 30, 2023. For this promotion there is no prior agreement or commitment of movements by the user against APM Terminals Califacture. case, the price of subsection 2.4.1.1 of the Tariff List is modified, the discounts of USD 2, USD 4 or USD 6 per ton apply to the modified price and in force at the time of invoicing, See invoicing details in the in the Regulation of Tariffs and Commercial Policy of APM Terminals Callao.

## (10) OFFER IN EXTENSION OF FREE DAYS FOR JOINT MOVEMENT OF BREAK BULK CARGO AND DRY CONTAINERS IN APM TERMINALS CALLAO

Dest coatomers and users.

APIA Terminos Callos (APIATC) has set an offer, which involves an extension of days free of use of the operating seas, for Those users who move break bulk cargo and dy containers by the DT 3014. This pomotion is governed by the following terms:

- Target group; Users entering both these bulk cargo and dy containers into APIAT Terminos Callos Temporary Depos (DT 3014)

- Temporary offers who are with annually more through DT 3014, minimum valueurs of 3 houseand two of the depositing seas for the user of the operating seas for the breakbulk cargo, which will be in addition to those 10 days of use of the operating seas provided by the integral callos (apic the user of the operating seas for the breakbulk cargo, which will be in addition to those 10 days of use of the operating seas provided by the integral callos (apic the user will have 30 days of our callos cargo and the callos (apic the user will have 30 days of our callos cargo and the season of the callos (apic the user will have 30 days of our callos cargo and the callos (apic the user will have 30 days of our callos cargo and the callos (apic the callos cargo and the callos of the callos (apic the callos cargo and the callos of the callos (apic the callos cargo and the callos of the callos (apic the callos cargo and the callos of the callos and the callos of the callos of

Effective The uses are spay or request this order from November 92, 2022 and June 3, 2022.

Requestimes the assess are part of the found the found the found the found the posting on December 1, 2022, then the accounting will be made until November 92, 2023, For this production of the found the posting on December 1, 2022, then the accounting will be made until November 92, 2023, For this production is a complete and as a hypothetical case. If the uses it is request for promotion is accepted and it is agreed that the posting on December 1, 2022, then the accounting will be made until November 92, 2023, For this production is the found to the found the posting on December 1, 2022, then the accounting will be made until November 92, 2023, For this production to the found the fou

## (11) DISCOUNT OR REDUCTION IN PRICE FOR LIVIAN VEHICLES IN THE TEMPORAL DEPOSIT OF APM TERMINALS CALL AG

Dear contenses and users.

APAT Terminist Collago (PATTC) has set discounts for the special service "Integrated Re-Ric Cargo Temporary Deptot Service" and for floors e-no cargo that request the service package that includes the standard ro-no cargo service (Cargo Portion) and the user of operating area (maximum) until day 10. These discounts are governed by the following terms:

- Target group; Users who were lefty inviterious for the Temporary Deptor of the Temporary Deptor Deptor of the Temporary Deptor of the Temporary Deptor Deptor of the Temporary Deptor Deptor of the Temporary Deptor Organization of the Service Service of the Temporary Deptor Dep . This promotion that implies a discount on

## (12) DISCOUNT OR ADDITIONAL REDUCTION IN TARIFF FOR LIQUID BULK CARGO - VEGETABLE OILS

APM Terminals Callao (APMTC) has arranged that, for vegetable oils that are unloaded in the form of liquid bulk cargo, a discount will be applied for the standard liquid bulk service (subsection 5.1.2.1 of the Tariff Schedule) if meet certain

(i) Discount of USD 0.5 per ton on the maximum liquid bulk standard service tariff (in force at the time of invoicing) for vegetable oil discharged through pipelines (no use of lecterns) This offer will be valid from January 1 to June 30, 2023.

## Published and updated on 21/12/2022

Applicable as from 01/01/2023
The special offers (reductions) of the regulated tariffs, unless otherwise expressly indicated in the Tariff List, will be in force until June 30, 2023.

Please visit our website: https://www.apmterminals.com/en/callao
For a greater detail of the scope and description of our services, commercial policies and surcharges, see the Regulation of Tariffs and Commercial Policy of APM Terminals Callao S.A.: \_https://www.